

LJMU Research Online

Girdhar, A, Harrison, CM, Mainieri, V, Bittner, A, Costa, T, Kharb, P, Mukherjee, D, Battaia, FA, Alexander, DM, Rivera, GC, Circosta, C, De Breuck, C, Edge, AC, Farina, EP, Kakkad, D, Lansbury, GB, Molyneux, SJ, Mullaney, JR, Silpa, S, Thomson, AP and Ward, SR

Quasar feedback survey: multiphase outflows, turbulence, and evidence for feedback caused by low power radio jets inclined into the galaxy disc

http://researchonline.ljmu.ac.uk/id/eprint/19850/

Article

Citation (please note it is advisable to refer to the publisher's version if you intend to cite from this work)

Girdhar, A, Harrison, CM, Mainieri, V, Bittner, A, Costa, T, Kharb, P, Mukherjee, D, Battaia, FA, Alexander, DM, Rivera, GC, Circosta, C, De Breuck, C, Edge, AC, Farina, EP, Kakkad, D, Lansbury, GB, Molyneux, SJ, Mullanev. JR. Silba. S. Thomson. AP and Ward. SR (2022) Quasar feedback

LJMU has developed LJMU Research Online for users to access the research output of the University more effectively. Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in LJMU Research Online to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.

The version presented here may differ from the published version or from the version of the record. Please see the repository URL above for details on accessing the published version and note that access may require a subscription.

For more information please contact researchonline@ljmu.ac.uk

http://researchonline.ljmu.ac.uk/

http://researchonline.ljmu.ac.uk/

Quasar feedback survey: multiphase outflows, turbulence, and evidence for feedback caused by low power radio jets inclined into the galaxy disc

A. Girdhar[®],^{1,2}★ C. M. Harrison[®],³★ V. Mainieri,¹ A. Bittner,^{1,2} T. Costa,⁴ P. Kharb[®],⁵ D. Mukherjee[®],⁶ F. Arrigoni Battaia[®],⁴ D. M. Alexander,⁷ G. Calistro Rivera[®],¹ C. Circosta,⁸ C. De Breuck,¹ A. C. Edge,⁷ E. P. Farina[®],⁴ D. Kakkad,^{9,10} G. B. Lansbury[®],¹ S. J. Molyneux[®],^{1,11} J. R. Mullaney,¹² Silpa S.,⁵ A. P. Thomson¹³ and S. R. Ward^{1,2}

¹European Southern Observatory, Karl–Schwarzschild–Straße 2, D-85748 Garching bei München, Germany

- ⁸Department of Physics & Astronomy, University College London, Gower Street, London, WC1E 6BT, UK
- ⁹European Southern Observatory, Alonso de Cordova, 3107, Vitacura Casilla 19001, Santiago, Chile
- ¹⁰Department of Physics, University of Oxford, Denys Wilkinson Building, Keble Road, Oxford OX1 3RH, UK
- ¹¹Astrophysics Research Institute, Liverpool John Moores University, 146 Brownlow Hill, Liverpool L3 5RF, UK
- ¹²Department of Physics and Astronomy, The University of Sheffield, Hounsfield Road, Sheffield, S3 7RH, UK
- ¹³ Jodrell Bank Centre for Astrophysics, Department of Physics & Astronomy, The Alan Turing Building, Upper Brook Street, Manchester M13 9PL, UK

Accepted 2021 December 29. Received 2021 December 20; in original form 2021 October 27

ABSTRACT

We present a study of a luminous, z = 0.15, type-2 quasar ($L_{[OIII]} = 10^{42.8}$ erg s⁻¹) from the Quasar Feedback Survey. It is classified as 'radio-quiet' ($L_{1.4\,GHz} = 10^{23.8} \text{ W Hz}^{-1}$); however, radio imaging reveals ~ 1 kpc low-power radio jets ($P_{jet} = 10^{44} \text{ erg s}^{-1}$) inclined into the plane of the galaxy disc. We combine MUSE and ALMA observations to map stellar kinematics and ionized and molecular gas properties. The jets are seen to drive galaxy-wide bi-conical turbulent outflows, reaching $W_{80} = 1000 - 1300 \text{ km s}^{-1}$, in the ionized phase (traced via optical emission lines), which also have increased electron densities compared to the quiescent gas. The turbulent gas is driven perpendicular to the jet axis and is escaping along the galaxy minor axis, reaching 7.5 kpc on both sides. Traced via CO(3–2) emission, the turbulent material in molecular gas phase is one-third as spatially extended and has three times lower velocity-dispersion as compared to ionized gas. The jets are seen to be strongly interacting with the interstellar medium (ISM) through enhanced ionized emission and disturbed/depleted molecular gas at the jet termini. We see further evidence for jet-induced feedback through significantly higher stellar velocity-dispersion aligned, and co-spatial with, the jet axis (< 5 °). We discuss possible negative and positive feedback scenarios arising due to the interaction of the low-power jets with the ISM in the context of recent jet–ISM interaction simulations, which qualitatively agree with our observations. We discuss how jet-induced feedback could be an important feedback mechanism even in bolometrically luminous 'radio-quiet' quasars.

Key words: galaxies: active - galaxies: evolution - galaxies: jets - quasars: supermassive black holes.

1 INTRODUCTION

Active galactic nuclei (AGNs) are the sites of growing supermassive black holes at the centre of galaxies (Kormendy & Ho 2013). They can produce enough energy, through accretion of matter, to exceed the binding energy of their host galaxies (Cattaneo & Best 2009; Bower, Benson & Crain 2012). With this tremendous amount of energy available, if it can efficiently couple to the gas extending from the vicinity of the AGN to galactic scales, the AGN could cause significant impact (known as 'feedback') on their host galaxies by either facilitating or suppressing the star-formation (see e.g. Binney & Tabor 1995; Ciotti & Ostriker 1997; Alexander & Hickox 2012; Fabian 2012; Harrison 2017). This 'AGN feedback' has become an imperative ingredient in cosmological galaxy formation simulations for them to reproduce key observables of galaxy populations and intergalactic material (e.g. Bower et al. 2006; McCarthy et al. 2010; Gaspari et al. 2011; Hirschmann et al. 2014; Vogelsberger et al. 2014; Henriques et al. 2015; Schaye et al. 2015; Taylor & Kobayashi 2015; Choi et al. 2018). However, understanding how this process occurs in the real Universe, particularly in the case of the most powerful AGN (i.e. quasars; $L_{bol} \ge 10^{45} \text{ erg s}^{-1}$), remains a challenge of extragalactic research. © 2022 The Author(s)

²Ludwig-Maximilians-Universität, Professor-Huber-Platz 2, D-80539 München, Germany

³School of Mathematics, Statistics and Physics, Newcastle University, NE1 7RU, UK

⁴Max Planck Institut für Astrophysik, Karl–Schwarzschild–Straße 1, D-85748, Garching bei München, Germany

⁵National Centre for Radio Astrophysics – Tata Institute of Fundamental Research, Pune University Campus, Post Bag 3, Ganeshkhind, Pune 411007, India ⁶Inter-University Centre for Astronomy and Astrophysics, Post Bag 4, Pune 411007, India

⁷Centre for Extragalactic Astronomy, Department of Physics, Durham University, South Road, Durham DH1 3LE, UK

^{*} E-mail: Aishwarya.Girdhar@eso.org (AG);

Christopher.Harrison@newcastle.ac.uk (CMH)

One way that AGNs are known to be capable of interacting with their host galaxy's multiphase interstellar medium (ISM) is through driving galaxy-wide (> 0.1 - 10 kpc) energetic outflows and turbulences (e.g. Nesvadba et al. 2011; Liu et al. 2013; Cicone et al. 2018; Davies et al. 2020; Veilleux et al. 2020; Fluetsch et al. 2021; Roy et al. 2021; Venturi et al. 2021). Theoretically, these outflows may be driven by radiation pressure, jets or AGNdriven winds and could occur over scales from the accretion disc $(< 10^{-2} \text{ pc})$ through to galaxy-wide scales, where they could affect the properties of the gas in the ISM and beyond (Costa et al. 2018; Ishibashi, Fabian & Maiolino 2018; Mukherjee et al. 2018b; Costa, Pakmor & Springel 2020; Morganti et al. 2021). However, from an observational perspective, whilst these multiphase outflows are now seen to be common in quasar host galaxies, controversy remains over the relative role of the different possible driving mechanisms (Hwang et al. 2018; Wylezalek & Morganti 2018; Jarvis et al. 2019; Somalwar et al. 2020). In particular in this work, we are interested in the relative importance of radio jets in driving multiphase outflows (see e.g. Morganti, Oosterloo & Tadhunter 2005; Nesvadba et al. 2006, 2008; Mullaney et al. 2013; Jarvis et al. 2019; Molyneux, Harrison & Jarvis 2019; Liao & Gu 2020; Venturi et al. 2021, Ramos Almeida et al.2021).

Also from a theoretical perspective, AGN can regulate black hole growth and star formation (i.e. causing a 'negative feedback' effect) by expelling gas through outflows, preventing gas from cooling and/or destroying star forming ISM (e.g. Silk & Rees 1998; Benson et al. 2003; Granato et al. 2004; Hopkins et al. 2006; Booth & Schaye 2010; Faucher-Giguère & Quataert 2012; King & Pounds 2015; Costa et al. 2018, 2020). On the contrary, AGN may also induce a 'positive feedback' effect due to the compression of the gas from outflows or jets, which induce localized enhancement of star formation (e.g. Zubovas et al. 2013; Ishibashi, Fabian & Canning 2013; Fragile et al. 2017; Lacy et al. 2017; Wang & Loeb 2018; Gallagher et al. 2019). However, from an observational perspective, the impact of the most luminous AGN on their host galaxies' ISM and star-forming properties is a matter of ongoing debate (e.g. Husemann et al. 2016; Villar-Martín et al. 2016; Maiolino et al. 2017; Harrison 2017; Kakkad et al. 2017; Cresci & Maiolino 2018; Perna et al. 2018; Rosario et al. 2018; do Nascimento et al. 2019; Bluck et al. 2020; Yesuf & Ho 2020: Scholtz et al. 2021).

Understanding the drivers of outflows and the impact of AGN on their host galaxies requires multiwavelength observations. Since outflows are multiphase, different diagnostics are required to study their properties and impact on their host galaxy. For example, spatially resolved observations in a small number of systems observed with both integral-field spectroscopy (IFS) and interferometers have shown that the ionized gas and molecular gas kinematics could be completely decoupled (Shih & Rupke 2010; Sun et al. 2014). This further highlights the need to study multiple gas phases at comparable resolutions to fully understand the impact of the feedback. In this work, we will trace warm ($\sim 10^4$ K) ionized gas kinematics using optical emission-lines, mostly [O III], which has long been used as a diagnostic to search for non-gravitational gas kinematics in AGN host galaxies (e.g. Weedman 1970; Stockton 1976; Heckman et al. 1981; Vrtilek 1985; Boroson & Green 1992; Veilleux et al. 1995; Harrison et al. 2014; Jarvis et al. 2019; Scholtz et al. 2021). For tracing cold (i.e. tens of Kelvin) molecular gas, following many other works, we will make use of the CO emission line, specifically the ¹²COJ = 3-2 transition as a tracer (e.g. Combes et al. 2014; Sun et al. 2014; Cicone et al. 2021; Circosta et al. 2021).

This work is part of the Quasar Feedback Survey (QFeedS; Fig. 1; Jarvis et al. 2021). This survey aims to address the challenges outlined



Figure 1. [O III] FWHM versus $L_{\rm [O III]}$ for z < 0.2, spectroscopically selected AGN from Mullaney et al. (2013), represented as blue data points and density contours. Stars represent the 42 QFeedS targets, with $L_{\rm [O III]} > 10^{42.1} \, {\rm erg \, s^{-1}}$ (dashed line), from which the quasar of this study was selected (highlighted with a black circle). The QFeedS targets are colour-coded by their radio luminosities ($L_{1.4\rm GHz}$). The quasar in this study has a modest, representative radio luminosity ($L_{1.4\rm GHz} = 10^{23.8} \, {\rm W \, Hz^{-1}$), and it is classified as 'radio-quiet' using traditional criteria (see Section 2.1).

above by studying the spatially resolved multiwavelength properties of 42 relatively low redshift (z < 0.2) guasar host galaxies. Using this redshift range allows us to obtain sensitive, high spatial resolution observations (typically ≤ 1 kpc), whilst also yielding a reasonable sample of powerful quasars ($L_{
m bol}\gtrsim 10^{45}\,{
m erg\,s^{-1}}$) representative of L_{\star} at the peak cosmic epoch of growth, where quasar feedback is expected to dominate (Kormendy & Ho 2013). Several pilot studies have been conducted on a subset of the QFeedS targets (Harrison et al. 2014, 2015; Lansbury et al. 2018; Jarvis et al. 2019, 2020), whilst an overview of the full sample can be found in Jarvis et al. (2021). Spatially resolved radio observations form the basis of QFeedS, which provides important insights into the prevalence and properties of radio jets in what is a representative and predominantly 'radioquiet' quasar sample (Jarvis et al. 2019, 2021). Here, we define radio-quiet using the criterion of Xu, Livio & Baum (1999) that compares the radio to [O III] emission-line luminosity. For a detailed discussion in the context of the QFeedS sample, see Jarvis et al. (2021).

In this work, we study one target from QFeedS for which we combine published radio observations with new observations from the integral field spectrograph Multi Unit Spectroscopic Explorer (MUSE; Bacon et al. 2010, 2017) and from the Atacama Large Millimeter/submillimeter Array (ALMA). The former enables us to map the stellar kinematics and ionized gas properties and the latter enables us to map the molecular gas properties. In Section 2, we introduce the target, observations, and data reduction. In Section 3, we estimate radio jet properties from existing data. In Section 4, we describe the approaches employed to fit the spectra and map the stellar kinematics and gas properties. In Section 5 and 6, we present our results and discussion, respectively. Finally, in Section 7, we summarize our conclusions.

We have adopted the cosmological parameters to be $H_0 = 70$ km s⁻¹ Mpc⁻¹, $\Omega_M = 0.3$, and $\Omega_{\Lambda} = 0.7$, throughout. In this cosmology, 1 arcsec corresponds to 2.615 kpc for the redshift of z = 0.15 (i.e.



Figure 2. A multiwavelength and multiscale view of our target galaxy. Centre: An optical continuum image from the MUSE data showing the local, 24×24 arcsec ($\sim 62 \times 62$ kpc), environment of J1316+1753. The primary target galaxy is centred at a position (0,0) with two clear spiral arms. To the north-west is an associated companion galaxy and to the south one passive satellite galaxy can be seen. Right: A closer look (18×16 arcsec; $\sim 47 \times 42$ kpc) showing the pseudo narrow-band [O III] image (from MUSE) with CO (3-2) emission (from ALMA) overlaid as contours in purple (with levels [16,8,4,2,1] RMS_{CO(3-2)}). The three colours are obtained by dividing the [O III] emission-line profile in three velocity slices as indicated in the legend. Left: A zoom-in of the continuum image in the central regions of the galaxy (4×4 arcsec; $\sim 10 \times 10$ kpc). The black contours are the VLA 6 GHz radio emission, at the levels [128,64,32] RMS_{radio}. In light green, we show three dots to highlight the positions of the radio core (HR:A) and the two radio jet hotspots (HR:B and HR:C), as defined in Jarvis et al. (2019). The central panel also shows the 'galaxy scale' (white inset; 10×10 arcsec; $\sim 26 \times 26$ kpc) within which the kinematic analysis is done. North is up and East is left in all panels and for all plots throughout this paper.

the redshift of the galaxy studied here). We define the radio spectral index, α , using $S_{\nu} \propto \nu^{\alpha}$.

2 TARGET, OBSERVATIONS, AND DATA REDUCTION

In Section 2.1, we describe the target investigated in this study in the context of the parent Quasar Feedback Survey and summarize its basic radio properties. We describe the details of the observations and data reduction for the MUSE and ALMA data in Sections 2.2 and 2.3, respectively. A summary of the data used throughout this work is presented in Fig. 2, with the central panel showing an optical continuum image from the MUSE data (\sim 62 × 62 kpc); the righthand panel showing a closer look (\sim 10 × 10 kpc) of the pseudo narrow-band [O III] image (from MUSE) with CO (3–2) emission (from ALMA) overlaid as contours in purple; and the left panel with a zoom-in (\sim 26 × 26 kpc) of the continuum image, focused on the central regions of the galaxy. A description of how these images were created is described in the following sub-sections and Section 4.1.

2.1 J1316+1753 in the context of the Quasar Feedback Survey

This paper focuses on a single target, J1316+1753, with an optical right ascension (RA) of $13^{h}16^{m}42^{s}90$ and declination (Dec.) of $+17^{\circ}53'32''_{.5}$ (J2000, ICRS). It is selected from QFeedS and this target illustrates the rich information gained on outflows and AGN feedback from our high-resolution multi-wavelength QFeedS dataset, comprising of radio, optical and sub-mm data. QFeedS was originally constructed using the emission-line selected AGN from Mullaney et al. (2013). Fig. 1 presents the 16,680 z < 0.2 AGN from Mullaney et al. (2013) in the [O III] full width at half-maximum (FWHM_{Avg}) versus [O III] luminosity parameter space. The QFeedS targets are indicated with stars in the figure. We note that the FWHM_{Avg} values plotted in Fig. 1 are the flux-weighted average of the FWHMs of the two emission-line components fitted by Mullaney et al. (2013). QFeedS targets were selected to have

quasar-level [O III] luminosities (i.e. $L_{[O III]} > 10^{42.1}$ erg s⁻¹; Jarvis et al. 2021). As described in Jarvis et al. (2021), a radio luminosity selection of $L_{1.4GHz} > 10^{23.45}$ W Hz⁻¹ is also applied to obtain the final QFeedS sample of 42 targets. This radio cut was set to be above the NVSS detection limit (Mullaney et al. 2013; Molyneux et al. 2019). Despite their moderate radio luminosities, the QFeedS sample still consists of 88 per cent 'radio-quiet' sources, based on the criteria of Xu et al. (1999) (see Jarvis et al. 2021), which is consistent with the 'radio-quiet' fraction of the overall quasar population (i.e. ~90 per cent; Zakamska et al. 2004). Furthermore, the QFeedS targets cover a representative range of emission-line widths of the parent population.

The target of this study, J1316+1753, is highlighted with a circle in Fig. 1. A comprehensive summary of the target's properties and its previous studies can be found in the supplementary material of Jarvis et al. (2021). However, briefly, J1316+1753 has a type-2 spectra; it has an optical emission-line redshift of z = 0.150 (see Jarvis et al. 2019); it is one of the most optically luminous sources in the QFeedS sample (with $L_{[OIII]} = 10^{42.8} \text{ erg s}^{-1}$); and it exhibits a broad [O III] emission-line profile (FWHM_{Avg} = 1022 km s^{-1}). The one-dimensional [O III] emission-line profile consists of two bright narrow emission-line components, in addition to a strong broad emission-line component (see Jarvis et al. 2019) and we investigate this emission using our spatially resolved MUSE data in Section 4. Based on the spectral energy distribution (SED) fitting from the UV to FIR, J1316+1753 is a star-forming galaxy with star formation rate = $30\pm10\,M_{\odot}\,yr^{-1}$, and stellar mass = $10^{11}\,M_{\odot}$ (see Jarvis et al. 2019 for these estimates).

Importantly, J1316+1753 has a modest radio luminosity of $L_{1.4GHz} = 10^{23.8}$ W Hz⁻¹ and, although it is confirmed to host a radio AGN, it is classified as 'radio-quiet', as per the traditional criteria (Xu et al. 1999; Best & Heckman 2012; Jarvis et al. 2021). Compared to some of the other QFeedS sources, J1316+1753 has relatively simple radio morphology (Jarvis et al. 2019, 2020). The 6 GHz radio image of this target from the Very Large Array (VLA) data was produced by Jarvis et al. (2019) (with a synthesized beam of

 0.33×0.28 arcsec; root-mean-square (RMS) noise of 8 µJy beam⁻¹) and the morphology reveals three components, which are most likely attributed to a radio core and two hotspots associated with radio jets. These were labelled as HR:A, HR:B and, HR:C in Jarvis et al. (2019), respectively, and we continue with this nomenclature in this work. The hot spots HR:B and HR:C are located at a projected distances of 0.91 kpc and 1.40 kpc from the central component, respectively. The central radio component, HR:A, has a flatter 1.5– 7 GHz spectral index of $\alpha \sim -0.3$, compared to the two hot spots, which have $\alpha \sim -1$ and $\alpha \sim -1.2$ (for HR:B and HR:C, respectively; Jarvis et al. 2019). Overall, the properties of J1316+1753 provide a cogent case to investigate the impact of low luminosity and compact radio jets on the multi-phase ISM in a 'radio-quiet' quasar host galaxy.

2.2 Observation and reduction of the MUSE data

J1316+1753 was observed with the Multi Unit Spectroscopic Explorer (MUSE; Bacon et al. 2010) on ESO's Very Large Telescope (VLT) as part of proposal ID: 0103.B-0071 (PI: Harrison). MUSE is an optical and near-IR integral-field spectrograph at the VLT, housed at the European Southern Observatory's Paranal Observatory. The observations were carried out in wide field mode, with adaptive optics, in a single observing block on the night of 2019 April 29–30. The observations consisted of 4 science exposures (of 550 s each) and 1 sky exposure (of 100 s), rotated by 90° and dithered within a ~4 arcsec box. The spectra ranges from 4750 to 9350 Å with a spaxel size of about 1.25 Å. The field of view is about 1 × 1 arcmin, with a pixel size of 0.2 arcsec.

The raw data were reduced using the standard ESO MUSE pipeline version 2.6 (Weilbacher et al. 2014, 2020). The flow of the pipeline includes bias-subtraction, correction for flat-field, illumination and geometry, and calibration in wavelength and flux. The sky emission was estimated using the sky exposure and subtracted from the science data. The datacubes were resampled on to a 0.2 arcsec \times 0.2 arcsec \times 1.25 Å grid. For the flux scaling, the relative offsets between the four exposures were computed using an average of 12 sources in the white-light images of each datacube. The final datacube was obtained by average-combining the four exposures corrected for their offset.

To estimate the seeing point spread function (PSF) for our cube, we used two point-like sources in our field of view. We estimated the PSF to be $\sim 0.75 \, \mathrm{arcsec}$, in the wavelength range around the [O III]₅₀₀₇ emission, using the MoffatModel2 function from the mpdaf package (Bacon et al. 2016). We corrected for the wavelength-dependent line-spread function (LSF), introduced by MUSE, during the line-profile fitting procedure, as described in Section 4.4.

2.3 Observation and reduction of the ALMA data

J1316+1753 was observed with ALMA in Band 7 in 3 × 1 h epochs between 2019 April 29–2019 May 03 under ALMA project 2018.1.01767.S (PI: A. P. Thomson). Our chosen correlator set-up comprises of three spectral windows, with one having a central frequency $v_{obs} = 345.795990$ GHz, corresponding to the redshifted CO(3–2) line at $z_{opt} = 0.15$ (Jarvis et al. 2019), and the others overlapping this spectral window with central frequencies of 344.7 and 347.0 GHz used for baseline subtraction. Each of the three spectral windows has a bandwidth of 1875 MHz (1870 km s⁻¹) and a spectral resolution of 31.25 MHz (31.17 km s⁻¹), recorded in dual polarization (XX, YY). In total, our three overlapping spectral windows provide 4.17 GHz (~3600 km s⁻¹) of spectral coverage

centred on the CO(3–2) line. Our science goals of mapping the galaxy-scale molecular gas kinematics and looking for jet-ISM interactions in the region of the radio jet, lead to using the C43-4 array, providing an angular resolution of $\theta_{res} \sim 0.5$ arcsec and a largest angular scale $\theta_{LAS} \sim 4$ arcsec (corresponding to linear scales of ~1.3 kpc and ~11 kpc, respectively, at the source redshift z = 0.15).

We obtained the raw data from the ALMA science archive and pipelined the data in CASA V5.4.0-68 using the reduction script supplied by ALMA user support. Following the successful completion of the ALMA CASA pipeline, the phases and amplitudes were visually inspected as functions of frequency and time using the CASA tool PLOTMS, during which a small percentage ($\lesssim 1$ per cent) of outliers were flagged from the calibrated data.

The reduced uv data were imaged in CASA TCLEAN. A spectral line datacube was created, using natural weighting (ROBUST = 2.0), onthe-fly primary beam correction (PBCOR = True) and multiscale clean at a pixel scale of 0.05 arcsec pix⁻¹, across an image of ~51 arcsec, (or 135 kpc) on a side. The resulting naturally-weighted images have a synthesized beam which well-resembles a two-dimensional Gaussian of width 0.54 arcsec × 0.45 arcsec at a position angle of 8°, with an rms sensitivity of $\sigma_{\rm rms} = 347 \,\mu Jy \, {\rm beam}^{-1} \, {\rm channel}^{-1}$.

2.4 Alignment of the MUSE, ALMA, and VLA data

To perform a multiwavelength analysis, we need to be sure that all the data are registered on the same astrometric solution. To do this, we followed the same procedure as in Jarvis et al. (2019), who demonstrated that the astrometric solutions of Sloan Digital Sky Survey (SDSS) images were accurate enough to align datacubes to VLA data, for 9 targets from QFeedS including J1316+1753, with a median accuracy of 0.13 arcsec (i.e. ~0.3 kpc). This is sufficient for the multiwavelength comparison that we perform in this work (Section 4). Briefly, we created pseudo broad-band images from the MUSE IFS cubes over the same wavelength range as the gband images from SDSS. The peak emission in the pseudo-broadband image was identified to be located 0.6706 arcsec from the peak radio emission and 0.6709 arcsec from the peak CO(3-2) emission (see Fig. 2; Section 4.1). Then we used the find_peaks function of the photutils (Bradlev et al. 2020) PYTHON package to identify the pixel associated to the peak emission in the MUSE pixel and anchored the coordinate values of this pixel to match the associated position from the image peak in SDSS. This resulted in a global astrometric shift of \sim 2.5 arcsec to the MUSE cube.

Since, after this shift, the peak continuum and the radio core were well aligned (i.e. within ~ 0.1 arcsec) this provides confidence in the alignment process, although we note that a priori these different emissions may not by physically aligned. Ultimately we chose the HR:A position of the radio core (introduced in Section 2.1; RA of $13^{h}16^{m}42^{s}8$ and Dec. of $+17^{\circ}53'32''_{...}3$) as our 'nucleus' position, indicated with a (0,0) in all relevant figures. All distances are measured with respect to this position as the reference point.

3 ESTIMATED JET PROPERTIES

In this section, we estimate the properties of the radio jets obtained from empirical relations and existing radio data for the target. These estimated properties are used to help interpret our new observations in terms of the effects of the jets on the quasar host galaxy in Section 6.2.1.

3.1 Jet kinetic power (P_{jet})

To estimate the kinetic jet power (P_{jet}) , we use the empirical relation of Merloni & Heinz (2007), obtained from a sample of low-luminosity radio galaxies, as below:

$$\log P_{\rm jet} = (0.81 \pm 0.11) \times \log L_{5\,\rm GHz} + 11.9^{+4.1}_{-4.4}.$$
 (1)

From the VLA data in Jarvis et al. (2019), the 5.2 GHz flux density of the radio core (HR:A) is 1.1 mJy (spectral index $\alpha = -0.3$). This corresponds to 5 GHz luminosity of $6.5 \times 10^{39} \text{ erg s}^{-1}$ for the radio core. Using the relation above we estimate a jet power of $7.8 \times 10^{43} \text{ erg s}^{-1}$. However, since the empirical relation is derived from lower resolution data than ours, we also calculate the jet power using the total 5 GHz flux density (i.e. over HR:A, HR:B, and HR:C), which is 2.7 mJy (spectral index $\alpha = -0.85$; Jarvis et al. 2019). Using this flux density, we estimate a jet power that is only higher by a factor of 2.5 (i.e. $1.7 \times 10^{44} \text{ erg s}^{-1}$). Hence, for our order-of-magnitude estimate, we assume the jet power to be, $P_{\text{jet}} = 10^{44} \text{ erg s}^{-1}$ during our discussion in Section 6.2.1.

3.2 Jet inclination angle (θ)

Since AGN jets experience bulk relativistic motions (Blandford & Königl 1979; Ghisellini et al. 1993), the apparent brightness of the different AGN sub-components can be used as an estimator of their inclination angle with respect to line of sight (los). The radio core prominence parameter is a measure of the core-to-extended radio flux density $[R_{\rm C} = S_{\rm core}/S_{\rm ext} = S_{\rm core}/(S_{\rm tot} - S_{\rm core})]$, and is a known statistical indicator of inclination angle, with respect to los, as discussed in detail in Orr & Browne 1982 (also see Saikia & Kapahi 1982; Kharb & Shastri 2004). We used the flux density estimates from the 5.2 GHz VLA data from Jarvis et al. 2019, which estimates the $S_{\text{core}} = 1.1 \text{ mJy}$ and $S_{\text{tot}} = 2.7 \text{ mJy}$. Using these values we estimated the $R_{\rm C}$ to be = 0.7, which gives $\log_{10}(R_{\rm C}) = -0.16$. The log $R_{\rm C}$ values are seen to vary from -4 to +4 for a sample of randomly oriented radio galaxies and blazars, corresponding to inclination angles from 90° to 0° (see Kharb & Shastri 2004). Therefore, the estimated $\log R_C \sim 0$, value for J1316+1753 corresponds to an inclination angle of $\sim 45^{\circ}$ for the jet in this source, with respect to our line of sight. Hence, based on these statistical arguments, a projected inclination of \sim 45 ° with respect to our los, is assumed for the jets in the rest of the paper. Since we estimated our galaxy's inclination angle to be 46° (see Section 4.2), we conclude that the jets are seen to be in the plane of the galaxy disc.

3.3 Jet speed (β)

The relativistic doppler beaming effects in jets enable us to estimate which is the approaching and which is the receding jet, and to also estimate a jet speed (Ghisellini et al. 1993; Urry & Padovani 1995). Using the 5.2 GHz VLA image from Jarvis et al. (2019), we measured the surface brightness (SB [μ Jy beam⁻¹]) values at equal distances from the peak surface brightness point (i.e. the radio core, HR:A) on both sides along the jet axis. We define the ratio of these surface brightness values as $J = SB_{\text{HR: B}}/SB_{\text{HR: C}}$, and repeated these measurements at seven different distances ranging from 0.07 to 1.2 arcsec. A range of *J* estimates was obtained from 1.1 to 2.6. On average, the south-eastern jet (HR:B) was measured with ~ 2 times higher surface brightness than the north-western jet (HR:C). Hence, it was concluded that HR:B is the approaching side of the jet.

Furthermore, the ratio, J was used to estimate the jet speed, $V_{jet} = \beta c$, using equation (A10) from Urry & Padovani (1995) as follows:

$$\beta \cos \theta = \frac{J^{1/p} - 1}{J^{1/p} + 1}.$$
 (2)

We assumed the value of p = 3.0 (as defined for a continuous, non-blobby jet, with $\alpha = -1$; Urry & Padovani 1995).¹ Using our estimate of the jet inclination angle ($\theta = 45^{\circ}$; see Section 3.2), we estimate the jet speed to be in the range of 0.04 c - 0.2 c. This seems to be a reasonable value, since for a typical low-luminosity AGN, jet speeds would be expected to be around 0.15 c (Ulvestad et al. 1999). A limitation of the jet speed and jet inclination estimates is that this relation is defined for radio-loud AGNs, in contrast to our radio-quiet source. However, we mainly use these values for a qualitative comparison to simulations and order-of-magnitude estimates in Section 6.2.1.

4 ANALYSIS

In this section, we describe our analysis methods to (1) produce the continuum and emission-line images shown in Fig. 2 (Section 4.1); (2) estimate the structural parameters of the galaxy (Section 4.2); (3) map the stellar kinematics using the MUSE data (Section 4.3); (4) model the optical and CO(3–2) emission-line profiles from the MUSE and ALMA data, respectively (Section 4.4); and (5) estimate the non-parametric properties of the emission-lines and produce their maps (Section 4.5). As illustrated in Fig. 2, we perform the analyses on two spatial scales, firstly over the whole 'galaxy-scale' (for which we map the stellar and gas properties using bins following Voronoi tesellation, see Section 4.3) and, secondly, focusing on the inner 'central-scale' region around the jets (for which we map the gas properties only).

4.1 Environment and central-regions of J1316+1753

We present pseudo-broadband and narrow-band (emission-line) images of J1316+1753 in Fig. 2. We note that for all images in this paper, North is up and East is to the left, as indicated in the central panel of this figure. The central panel shows the stellar continuum image of the galaxy inside a 24×24 arcsec (62×62 kpc) region. To make the image, we median-collapsed the data cube (masking all the emission-lines) across the same wavelength range as that used for the stellar continuum fitting (see Section 4.3 for a detailed explanation and Fig. 3 for a representation). The figure clearly reveals the galaxy morphology, with a central bulge and two prominent spiral arms extending over ~ 21 kpc. Within this image we identify, in addition to our main target, one passive satellite galaxy to the south (without any emission-lines). However, based on the regular morphology and kinematics (discussed in Section 5.2), we do not see any clear disturbances in the galaxy dynamics caused by it that could affect our conclusions and hence they are not discussed henceforth in this work. We make particular note of the bright and clumpy continuum structure located \sim 17 kpc to the north-west of the primary galaxy nucleus. For identification purposes only, we label this as the 'companion galaxy'. We note that this is identified in emission lines at the same redshift as our target galaxy, meaning that it is also

¹We note that Urry & Padovani (1995) use the opposite sign convention for spectral index (α) in their study compared to this analysis.



Figure 3. Examples of our emission-line and stellar continuum fits to the MUSE and ALMA data. Upper left: [O III] emission-line profile (grey curve) and our best fit (dashed black curve), as well as the three individual Gaussian components used to construct the total fit (yellow, blue, and orange curves). Upper right: CO(3-2) emission-line profile (grey curve), with the best fit (dashed black curve) and the two Gaussian components that make up the total fit (blue and red curves). In both upper panels, we indicate the non-parametric values obtained from our analyses: V_{50} (dashed green line) to trace the bulk velocity and W_{80} (green arrows) to trace the velocity-dispersion, which is the velocity width between V_{10} and V_{90} (vertical dashed grey lines). Both emission-line profiles are extracted from Voronoi Bin no. 2, as defined in Section 4.3. Lower panel: Spectra (black curve), presented across the wavelength range used to obtain the stellar continuum fits (red curve) for estimating the stellar kinematics (see Section 4.3). Wavelength regions excluded from the fits (due to strong emission-lines and the instrumental chip gap) are highlighted with shaded regions. The fit residuals are shown as green curves.

physically associated. The kinematic analysis in Section 5.2 is done within the region indicated through the white box. We refer to it as the 'galaxy-scale' region $(10 \times 10 \operatorname{arcsec} \operatorname{or} 26 \times 26 \operatorname{kpc})$.

The right-hand panel of Fig. 2 covers a spatial region of $18 \times 16 \operatorname{arcsec}$ (i.e. $47 \times 42 \operatorname{kpc}$), chosen to show the extent of the [O III] gas, we identified in and around the main galaxy. The three-colour image shown is comprised of three pseudo narrow-band [O III] images, collapsed over the velocity slices indicated in the legend. The overlaid contours in purple represent the CO(3–2) image, created by collapsing the ALMA cube over a frequency range of $\pm 700 \operatorname{km s^{-1}}$. The black overlaid contours are from the 6 GHz VLA image (see Section 2.1; Jarvis et al. 2019).

In the left-hand panel of Fig. 2, we show a zoom-in of the continuum image, as indicated by the yellow box in the main panel, extending over $\sim 4 \operatorname{arcsec}$ (i.e. 10 kpc). We use this region as the 'central-scale' to perform our spaxel-by-spaxel kinematic analyses in Section 5.3. The three light-green colour dots indicate the position of the radio core (HR:A) and the two jet hotspots (HR:B and HR:C; discussed in detail in Section 2.1). The cyan colour ellipse shows an isophote fit to the stellar bulge profile, and is discussed later in Section 4.2.

4.2 Position angle, inclination angle, and sizes

To aid the discussion on how the jets and gas interact, it is useful to constrain some basic properties of the host galaxy stellar emission, specifically the size, major axis position angle, and inclination angle. We used the high spatial resolution continuum image (Fig. 2; middle panel) to make measurements of half-light radii ($R_{1/2}$), morphological axes (PA), and inclination angle (θ). We fitted a two-dimensional Gaussian model to our galaxy's continuum image to obtain a morphological position angle = $152 \pm 2^{\circ}$, an axis ratios (b/a) = 0.71 ± 0.04 , and $R_{1/2} = 6.10 \pm 0.12$ kpc. We note that all quoted position angles are defined as going East of North. The half-light radius estimate was further verified by fitting an exponential Sersic profile to the surface brightness of the galaxy disc. Furthermore, following the works of Tully & Fisher (1977), Weijmans et al. (2014), we used the axis ratios (b/a) to estimate the inclination angle (θ) using the following relation:

$$\cos^2 \theta = \frac{(b/a)^2 - q_0^2}{1 - q_0^2},\tag{3}$$

where q_0 is the intrinsic axial ratio of an edge-on galaxy (Tully & Fisher 1977). We assumed $q_0 = 0.2$, which is appropriate for thick

discs, however, for our axial ratio values, a change in q_0 by a factor of 2 would result in a < 9 per cent change in the inclination angle (θ) . Using the above relation, we estimated a galaxy inclination angle of $46 \pm 2^\circ$. We would like to note that the reported uncertainties are purely from the fitting function and do not take into account the intrinsic uncertainties arising due to the choice of a simple model.

In Section 6.2.2, we are interested in the relative position angle between the radio jets and the observed stellar bulge. Therefore, we followed a similar approach for the stellar bulge (Fig. 2; left-hand panel) and fitted a 2D Gaussian (cyan ellipse) to obtain a position angle of $139 \pm 4^{\circ}$.

4.3 Stellar kinematics from MUSE

The gas kinematics in centres of galaxies can be a good tracer of the AGN-driven outflows and the consequent jet–ISM interactions. However, the effect of the underlying stellar kinematics needs to be quantified to de-couple gravitational-motions from gas dynamics arising due to other processes. We use the MUSE data to map the stellar kinematics over the full extent of the stellar emission, i.e. the 'galaxy-scale' indicated by the white box in Fig. 2.

We fitted the stellar continuum using the modular analysis framework of GIST pipeline² (Galaxy IFU Spectroscopy Tool; Bittner et al. 2019). As a brief overview, the GIST pipeline works directly on the science-ready MUSE datacubes, performs a Voronoi tesellation (Cappellari & Copin 2003), fits spectral templates to the observed spectra, and outputs a best-fitting estimate of stellar velocity and stellar velocity-dispersion for each Voronoi bin. A complete overview of the pipeline is presented in Bittner et al. (2019).

The spectra were de-redshifted to the rest frame using an initial estimate of the systemic redshift (z = 0.15). The spaxels below an isophote level with an average signal-to-noise ratio (SNR) of 5 per spectral bin, as measured within the rest-frame wavelength range of 4452–4630 Å (range devoid of any bright emission-lines), were excluded from the fits. Then, the remaining galaxy spaxels were spatially binned to achieve a minimum SNR using the adaptive Voronoi tesellation routine of Cappellari & Copin (2003). We set the minimum SNR per Voronoi bin to be 30, which resulted in stable measurements of the stellar kinematics, but still provided a good spatial sampling (see Section 5.2). As a result of this procedure, the galaxy was divided in 41 Voronoi bins over the galaxy-scale frame. We show the shape and location of these bins in the Appendix (Fig. A1). Of these, 38 bins (Bin No. 0-37) were contiguous and represented the main galaxy under investigation, while 3 bins (Bin No. 38, 39, 40) were associated with the minor companion galaxy (as discussed in Section 5.1, and indicated in Fig. 2).

The actual wavelength range used to fit the stellar continuum was restricted to 4185–5500 Å in the rest-frame. This wavelength range covers sufficient emission-line free regions of prominent stellar absorption features, whilst avoiding the redder wavelength range of MUSE that is more strongly affected by sky line residuals (also see e.g. Falcón-Barroso et al. 2006; Bittner et al. 2019).

The stellar continuum fitting with GIST requires the preparation of a spectral template library. We required a spectral template library that is compatible with sufficient wavelength range and the spectral resolution for our MUSE data. Following these requirements, we selected the second data release (DR2) of the X-Shooter Spectral Library (XSL; Arentsen et al. 2019; Gonneau et al. 2020). The XSL is a stellar spectral library covering the wavelength range from 3000

²http://ascl.net/1907.025

to 25000 Å, at a resolution of about $R \sim 10~000$. We used the ultra-violet to blue (UVB) set of the DR2 of xsL that covers 813 observations of 666 stars. The parametrization from Bacon et al. (2017) was adopted as the MUSE line spread function to broaden all the template spectra to the resolution of our observed spectra before any fits were conducted. This process corrects the velocity-dispersion measurements for the spectral resolution of MUSE.

The stellar kinematics were derived by performing a run of the pPXF routine (Cappellari & Emsellem 2004; Cappellari 2017). To account for small differences between the MUSE spectra and the templates, we included an additive Legendre polynomial of the order 8 (following literature, see Pinna et al. 2019; Gadotti et al. 2020; Bittner et al. 2020).³ Before the fitting, care was taken to mask wavelength ranges covered by the broad and bright emission lines. We adopted a conservative approach and used broad masks with a width of up to 200 Å for the broadest emission lines (e.g. Ne IV, [Ar IV], $H\beta$, [O III]₄₉₅₉, [O III]₅₀₀₇, etc.; see Fig. 3) in order to ensure that these lines were masked even for the central spaxels with the broadest emission lines. This ensured that no contamination from the emission lines affected the analysis of the stellar kinematics, but still left sufficient stellar absorption features to obtain stable solutions for the stellar kinematics. We note that on reducing the masking width to 100 Å for the strongest emission lines, the velocity measurements are consistent in the outer Voronoi bins where the strongest emission lines are much narrower, giving us confidence the fitting wavelength coverage is sufficient. In addition, the wavelength range from 5820 to 5970 Å is masked, to avoid the contamination of the spectra from the laser guide stars of the adaptive optics facility. The stellar fits were visually inspected for each bin and were seen to closely follow the observed spectra in each case. As an example illustration, the observed spectra and the fits to the stellar continuum are shown in the lower panel of Fig. 3 (see supplementary material for the fits from all other Voronoi bins). In Section 5, we discuss the resulting stellar kinematics. The median of errors, across all Voronoi bins, on the stellar kinematics were $\pm 16 \text{ km s}^{-1}$ for stellar velocities and \pm 18 km s⁻¹ for stellar dispersion estimates, where the errors are formal errors (1σ) uncertainties estimated from pPXF.

We used the redshift estimate from the stellar fits of each Voronoi bin and estimated a flux-weighted mean redshift of z = 0.15004. This value has then been used as the systemic redshift for all of the analysis in this paper.

4.4 Emission-line profile fitting procedure

In this work, we characterize the spatially resolved kinematics and flux ratios of strong optical emission-lines (using the MUSE data) and the CO(3-2) emission line (using the ALMA data). We take the same overall approach for fitting all emission lines, which we describe first, before giving details that are specific to fitting individual emission lines. All of the line profiles and velocity maps displayed in this work have been shifted to the rest frame, using the stellar systemic redshift described in Section 4.3. A set of emission-line profiles and fits from across the whole galaxy are shown in the supplementary material.

We fitted the profiles using the χ^2 minimization procedure CURVE_FIT (Virtanen et al. 2020) and the most appropriate num-

 $^{^{3}}$ We note that we tried the fit for two different cases, one with only an additive Legendre polynomial and the other with only a multiplicative Legendre polynomial (both with order 8). We find no qualitative difference in the kinematic estimates and only a moderate variation in absolute measured velocity values of about 6 km s⁻¹ (median value).

ber of Gaussian components (detailed below) were selected using the Akaike Information Criteria (AIC; Akaike 1974). Due to the complexities of the optical emission-line profiles, we primarily use these fits to obtain an overall non-parametric characterization of the line profiles (see Section 4.5) and largely refrain from assigning physical significance to individual components (following e.g. Liu et al. 2013; Harrison et al. 2014). However, we verified that the nonparametric results on the gas kinematics provide qualitatively the same conclusions as investigating 'broad' and 'narrow' components separately.

For our primary tracer of the ionized gas kinematics, we use the $[O \text{ III}]\lambda5007$ emission line. We fitted the spectral region around the $[O \text{ III}]\lambda\lambda4959,5007$ emission-line doublet using the rest-frame air wavelengths in the range [4940, 5040] Å. We attempted to fit the $[O \text{ III}]\lambda5007$ profile using one, two or three Gaussian components (selected based on the AIC). An example $[O \text{ III}]\lambda5007$ (hereafter [O III]) emission-line profile and the best-fitting solution is shown in Fig. 3. We used the same Gaussian component parameters to fit simultaneously the $[O \text{ III}]\lambda4959$ emission profile, by using the same line width, a fixed wavelength separation and a fixed intensity ratio of $[O \text{ III}]\lambda4959/[O \text{ III}]\lambda5007 = 0.33$ (e.g. Dimitrijević et al. 2007).

For the purpose of obtaining estimates of electron densities (see Appendix B), we followed a similar procedure to fit other optical emission-lines (in all cases using rest-frame air wavelengths). We note that we used a simple first-order polynomial to characterize the local continuum around all the optical emission lines. This is typically appropriate for the strong emission-line-dominated spectra that we have and hence will not affect the inferred ionized kinematics.

To trace the molecular gas kinematics, the CO(3–2) line (with a rest frame frequency of 345.7960 GHz) was extracted from the ALMA data cube around a spectral window of $\sim \pm 700$ km s⁻¹. Based on visual inspection and an AIC estimate, a maximum of two Gaussian components were required to characterize the CO(3–2) emission-line profile over the whole galaxy. The local baseline continuum was well characterized with a constant value. An example CO(3–2) emission-line profile and the best-fitting solution is shown in Fig. 3 and a set of emission-line profiles and fits from across the whole galaxy are shown in the supplementary material.

4.5 Non-parametric velocity definitions

The complexities and degeneracies of the emission-line fits exhibited in powerful quasars lead us to favour non-parametric definitions of bulk velocity shifts and velocity-dispersion (also see Harrison et al. 2014; Jarvis et al. 2019, for a kinematic study of other QFeedS targets). Therefore, we used the best-fit models of the emission-line profiles from Section 4.4 to calculate the following:

(i) V_{50} , which is the median velocity of the overall emission-line profile, and is a tracer of the bulk gas velocity.

(ii) W_{80} , which is the velocity width of the emission line that contains 80 per cent of the integrated flux and is a tracer of the velocity-dispersion. This is defined as, $W_{80} = V_{90} - V_{10}$, where V_{10} and V_{90} are the 10th and 90th percentiles of the emission-line profile, respectively. For a single Gaussian profile, W_{80} is equivalent to $1.083 \times FWHM$ or $2.6 \times \sigma$.

(iii) Peak SNR ratio, which is simply the ratio of the peak flux density in the emission-line profile to the noise in the spectra. The noise is calculated by taking the standard deviation of the continuum spectra, in windows of ~ 30 Å, on both sides of the emission peak.

We note that we corrected the W_{80} values by first subtracting, in quadrature, the line spread function (LSF) σ_{LSF} , from each of the component gaussians. The wavelength-dependent σ_{LSF} of the MUSE data was estimated using equation (8) from udf-10 parametrization of Bacon et al. (2017).

In addition to tracing these non-parametric values, we also attempted to trace the properties of the individual broad and narrow components. Although these results are more subject to degeneracies, and therefore provided noisier results, we found that we draw the same conclusions on the spatial distribution of the high velocitydispersion gas and positions of bulk velocity offsets discussed in Sections 5.2 and 5.3. This can be visually verified by looking at the individual Gaussian components of the fits shown in the supplementary material.

To map the ionized and molecular gas kinematics and the emissionline flux ratios on the galaxy-scale, we use the same spatial Voronoi bins as obtained for the stellar kinematics maps described in Section 4.3. This approach is sufficient for our goal of establishing a broad description of the galaxy kinematic structure and performing a like-for-like comparison between the stellar kinematics and the two gas phases (see Section 5.2). Furthermore, we find that the gas is predominantly located within the extent of the stellar emission (see Fig. 2; Section 5.1). We note that for the CO(3-2) gas, some bins had low SNR (i.e. < 3; see Fig. A), and are hence not shown for the ALMA maps. A study of any additional faint gas in the circumgalactic medium is deferred to future work. To plot the values extracted from the fits in the maps, on the galaxy-scale, we required a signal-to-noise ratio (SNR \geq 3). However, this only affected the galaxy-scale maps of CO(3-2) where the SNR was found to be lower than this threshold in three bins. Maps of V_{50} and W_{80} for [O III] and CO(3-2) are discussed in Sections 5.2 and 5.3. A visual inspection of the data reveals that our maps provide a reliable and meaningful representation of the underlying data.

We also produced maps of the emission-line properties on the central-scale around the jets (see yellow box in Fig. 2). For the central-scale maps, we performed the emission-line profile fits, following Section 4.4, in spatial regions of 0.2×0.2 arcsec. Again, we required an SNR ≥ 3 to plot the derived quantities in these maps. Central-scale maps of SNR, V_{50} and W_{80} for both [O III] and CO(3–2) are presented and discussed in Section 5.3. We note that we lack sufficient SNR to map the stellar kinematics, using these smaller spatial bins on this scale. For all the fitted parameters, we obtained the one standard deviation errors as $\sqrt{\text{diag}(\text{pcov})}$, where pcov is the estimated covariance of each parameter. To estimate the uncertainty in W_{80} , we used the error in the FWHM of the broadest Gaussian component and for the uncertainty in V_{50} we used the error in the velocity estimate of the narrowest Gaussian component.

5 RESULTS

In this section, we present our results from the analysis of MUSE and ALMA data for the z = 0.15, type-2 quasar, J1356 + 1753, selected from QFeedS (see Fig. 1). Specifically, we present (1) the distribution of stellar emission, ionized gas, and molecular gas (Section 5.1); (2) the galaxy-scale stellar and multiphase gas kinematics (Section 5.2); and (3) the central-scale gas kinematics and emission-line properties close to the jets (Section 5.3).

5.1 Overview of the stellar and gas distribution

Fig. 4 presents the galaxy-scale stellar kinematics (first column), ionized gas kinematics (second column), residuals by subtracting the stellar and ionized gas kinematics (third column), the molecular gas kinematics (fourth column), and the residuals from subtracting



Figure 4. A comparison of the stellar and multiphase galaxy-scale kinematics using the Voronoi bins. Top row: Bulk velocity maps (V_{\star} for the stellar and V_{50} for the gas). Bottom row: Velocity-dispersion maps (σ_{\star} for the stellar kinematics and W_{80} for the gas). In both rows, from left-to-right: stellar kinematics, ionized gas kinematics (from [O III]), residual ionized gas kinematics (i.e. the velocity difference between the stellar and ionized gas maps; see Section 5.2.2), molecular gas kinematics [from CO(3–2)], and residual molecular gas kinematics. In all panels, the light-green contours represent the 6 GHz radio emission at levels [32,16,4] RMS_{radio}. Grey contours (in second column) trace the [O III] emission at [64,16,4,2,1] RMS_[O III]. The dark-green contours trace the CO(3–2) emission at levels of [16, 8, 4, 2, 1] RMS_{CO(3–2)}. The dashed green and dotted pink lines trace the projected jets and stellar kinematic axis, respectively. A scale bar representing a length of 5 kpc is shown in both the right-most panels. The bulk of the gas follows the stellar velocities at the galaxy scale as seen in both the phases, except for the central bins, around the jet region, which show strong velocity residuals (top-row: third and fifth columns). The highest dispersion values are seen perpendicular to the jets for ionized gas (bottom-row, third column), indicated by the cyan lines (traced from the cones of highest velocity-dispersion identified in Fig. 5).

the molecular and stellar kinematics (fifth column). In all columns, the upper-row presents the bulk velocity maps (V_{\star} or V_{50}) and the lower row presents the velocity-dispersion maps (σ_{\star} or W_{80}). We note that while computing residuals, the stellar dispersion values were scaled by a factor of 2.6, to make the values more directly comparable to W_{80} , which we used to characterize the velocity width of the gas (see Section 4.5 for the estimation methods).

As can be seen in Fig. 2, J1316+1753 is a spiral galaxy extending over ~20 kpc ($R_{1/2} = 6.1$ kpc) from the north spiral arm to the south spiral arm. [O III] emission is our tracer of the ionized gas. The [O III] image shown in the left-most panel of the figure reveals that the ionized gas broadly covers the same spatial regions as the stellar continuum emission in both the primary galaxy and the minor companion galaxy to the north west (also see the second column of Fig. 4). However, we also observe an ionized gas cloud (i.e. with no corresponding stellar continuum emission), at a distance of ~ 17 kpc south east of the galaxy nucleus, with a blue-shifted velocity of -250 km s⁻¹. Although not the focus of this work, we speculate that this cloud may be a tidal debris or the result of an historic outflow (see Lintott et al. 2009; Keel et al. 2012, 2019 and the references therein).

CO(3-2) emission is our tracer of the molecular gas (see Section 2.3). The CO(3-2) contours shown in the right-most panel of Fig. 2 and in the fourth column of Fig. 4 reveal that the molecular gas broadly follows the distribution of the stellar continuum emission, including tracing the northern and southern spiral arms. However, at the sensitivity of our observations, the molecular gas is less extended than both the [O III] and continuum emission. Specifically, CO(3-2) emission is not detected towards the outer edges of the spiral arms. None the less, for this work we only aim to qualitatively describe the galaxy-scale molecular gas kinematics, for which the data is sufficient.

5.2 Kinematics on the galaxy scale

As a first investigation of the stellar and multiphase gas kinematics, we look at the entire galaxy scale (i.e. across a 26×26 kpc region; see Fig. 2), using the Voronoi bins defined in Section 4.3 (also see Fig. A1). By using the same spatial bins, we are able to directly compare the stellar, [O III] and CO(3–2) kinematics without having to rely on a particular dynamical model with the associated uncertain assumptions. We note that for the following results, we are only concerned with the contiguous Voronoi bins of the central galaxy and we exclude the minor companion galaxy when presenting quantitative values. Our goal here is to qualitatively assess if the multiphase gas follows the host galaxy's gravitational motions, before performing a deeper investigation of the kinematics and distribution of the gas in the central region of the galaxy in Section 5.3.

5.2.1 Stellar kinematics

The stellar velocity and stellar velocity-dispersion maps, presented in Fig. 4, clearly suggest the presence of a smoothly rotating stellar galaxy disc. We obtain an approximate major kinematic axis of 140° by connecting the Voronoi bins with the highest values of projected velocities (i.e. $-268 \pm 18 \text{ km s}^{-1}$ and $+240 \pm 14 \text{ km s}^{-1}$) through the galaxy nucleus. This major kinematic axis is well aligned with the morphological axis of 150° , which we measured in Section 4.2. Accounting for the 46° inclination angle of the galaxy (see Section 4.2), we estimated the de-projected maximum rotation velocities of -360 and $+330 \text{ km s}^{-1}$. These high rotation velocities are consistent with previous work, suggesting that this galaxy has a very high mass (i.e. the inferred stellar mass from SED fitting is = $10^{11 \pm 0.3} M_{\odot}$; Jarvis et al. 2020).

The observed stellar velocity-dispersion ranges from $81 \pm 22 \text{ km s}^{-1}$ in the outer spiral arms to much higher values between 290 and 360 km s⁻¹ in the very central regions, with a median value of $326 \pm 27 \text{ km s}^{-1}$ (see small inset in Fig. 4). We discuss the interesting origin of the highest central stellar velocity-dispersion, and the remarkable spatial alignment with the radio jets in Section 6.

5.2.2 Ionized gas kinematics

In Fig. 4 (second column), the kinematic structure of the bulk ionized gas velocity on large scales, i.e. $V_{50, [O \text{ III}]}$, can be seen to broadly follow the stellar velocity structure. Indeed, the residuals (third column) created by subtracting the stellar kinematic values from $V_{50, [O \text{ III}]}$, reveals only modest residuals in the majority of the bins outside of the central regions, with $|V_{50, [O \text{ III}]} - V_{\star}| \leq 50 \text{ km s}^{-1}$ in the spiral arms. We therefore infer that on large scales the [O III] emission is broadly following the gravitational motions of the host galaxy. In contrast, the residuals between $V_{50, [O \text{ III}]}$ and V_{\star} are much larger in the central, $\leq 10 \text{ kpc}$ regions of the galaxy, reaching values of up to -105 and $+260 \text{ km s}^{-1}$. We note that the bin at the tip of both spiral arms also shows a large velocity residual of $= 139 \text{ km s}^{-1}$ that we attribute to disturbed ionized gas extending as far as these locations (see below).

A comparison between the stellar velocity-dispersion and velocity-dispersion of the [O III] emission further confirms nongravitational ionized gas motions (Fig. 4). The $W_{80, [O III]}$ map reveals extreme ionized velocity widths over very large scales. With the exception of the outermost north west and south-east regions, the majority of the bins have [O III] velocity widths much greater than the local stellar velocity-dispersion. Most dramatically, perpendicular to the jet axis, high velocity-dispersion is seen with $W_{80} >> 2.6 \sigma_{\star}$ (with residuals of typically 740 km s^{-1}). The velocity-dispersion in these bins ranges from 1000 to $1300 \,\mathrm{km \, s^{-1}}$, and hence we identify this as the most disturbed gas with $W_{80} > 1000 \,\mathrm{km \, s^{-1}}$. This disturbed gas is seen to extend as far as the outer Voronoi bins that are located 7.5 kpc each side of the nucleus. While the disturbed gas may be extending beyond this distance, we take a conservative estimate of 7.5 kpc, obtained from the Voronoi bins defined at an SNR > 5 (as explained in Section 4.3). Therefore, we conclude that, whilst the [O III] gas follows the large-scale rotation of the host galaxy, we see significant non-gravitational motions around the jets that is detected to a projected distance of 7.5 kpc along the minor axis of the host galaxy.

For an estimate of the outflow velocities, we obtained the maximum (terminal) velocities following Heckman et al. (2004), as $V_{\text{max}} = \Delta V - FWHM_{\text{broad}}/2$, where $\Delta V = V_{\text{broad}} - V_{\star}$ (also see e.g. Rupke, Veilleux & Sanders 2005; Veilleux, Cecil & Bland-Hawthorn 2005). We found these velocity estimates⁴ in the terminal bins (precisely, Voronoi bin 28 and 33) to be similar to our non-parametric estimate of W_{80} ⁵ in the same location and hence we decided to use the W_{80} values for our estimate of the outflow velocity in Section 6.2.1 (also see e.g. Liu et al. 2013; Harrison et al. 2014). This analysis of the galaxy-scale ionized gas kinematics reveals non-gravitational motions in and around the nuclear radio jets in J1316+1753, with high velocity-dispersion gas extending to

```
{}^{4}V_{\text{max, Bin28}} = 480 \pm 17 \,\text{km s}^{-1}; V_{\text{max, Bin33}} = 650 \pm 10 \,\text{km s}^{-1}.
{}^{5}W_{80, \text{Bin28}} = 880 \pm 19 \,\text{km s}^{-1}; W_{80, \text{Bin33}} = 820 \pm 26 \,\text{km s}^{-1}.
```



Figure 5. Kinematic analysis of the ionized gas in the region around the central jet, using maps of the non-parametric values of the [O III] emission-line fits and position velocity diagrams around the [O III] emission line (see Section 5.3.1). The black contours in each map represent the 6 GHz radio emission at levels of [32,16,4] RMS_{radio}. The three cyan dots represent the HR:A (core) and jet hot spots (HR:B and HR:C), as in Fig. 2. A 1 kpc scale bar is shown in each of the maps. Panel (a): Peak-SNR map; **panel (b)** : W_{80} map; **panel (c)** : V_{50} map; **panel (d)** : PV diagram extracted along the jet axis (as shown through the dashed red axes in panel (a), with the position-offset along *x*-axis and velocity-offset along *y*-axis, and the vertical lines indicate the locations of the radio jet hot spots; **panel (e)** the [O III] line profile from regions of highest peak-SNR values; **panels (f) and (g)** : the [O III] line profiles from regions of disturbed gas, above (Voronoi bin 16) and below (Voronoi bin 7) the jet, respectively, and **panel (h)** : electron density map from [S II] doublet. The Gaussian components (in *panels, e, f, and g*) are as described in Fig. 3. In panel (b), the emission corresponding to an enhanced velocity-dispersion ($W_{80} \ge 1000 \text{ km s}^{-1}$) suggests a biconical structure in the region perpendicular to the jets (indicated by the dash-dotted cyan lines). The PV diagram in panel (d) reveals enhanced velocities behind the jet hotspots, as highlighted by the purple vertical arrows. Narrow velocity components, highlighted with horizontal dark green arrows in panel (d), are observed to propagate roughly from the radio hotspots and extend for 0.8 arcsec (2.1 kpc) in each direction.

a distance of 7.5 kpc each side of the nucleus. We investigate the velocity structure of the central regions in more detail in Section 5.3. We note that the spatially resolved emission-line ratio diagnostic for the dominant ionizing source reveals AGN to be the dominant ionizing source across all galaxy bins (except in the companion galaxy); however, we defer the detailed discussions to future works.

5.2.3 Molecular gas kinematics

Using the kinematics of CO(3–2) as a tracer in Fig. 4, we find that the molecular gas shows a similar, but less extreme, behaviour than the ionized gas. The bulk velocity of the molecular gas on large scales broadly follows that of the stellar velocity. The peak projected velocities of the CO(3–2) are similar to the peak stellar velocities and furthermore, the major kinematic axes are found to be around 140 ° for both the molecular and stellar kinematics. The $|V_{50, CO(3-2)} - V_{\star}|$ residuals are small (i.e. $\leq 70 \text{ km s}^{-1}$) in all but the central 6 Voronoi bins,⁶ which on average have residuals of 190 km s⁻¹. Therefore, the bulk velocity of the molecular gas is even closer to the stellar kinematics than found for the ionized gas, but still shows significant non-gravitational motion in regions around the radio jets. This is investigated further in Section 5.3.

Fig. 4 shows that, similar to the ionized gas, the molecular gas has a velocity-dispersion comparable to the stellar kinematics in the outer bins, but with much higher residuals (i.e. 110 km s⁻¹) in the central bins. However, whilst the central 7 Voronoi bins⁷ show the highest velocity widths of $W_{80, CO(3-2)} = 380-500 \text{ km s}^{-1}$, these are on average, a factor of 3 less than those seen in [O III] at the same location. The most distant spaxel where this disturbed molecular gas is observed (i.e. with $W_{80} > 380 \text{ km s}^{-1}$) in Fig. 4 has a projected distance from the radio core of 2.5 kpc. Therefore, the spatial extent of the high velocity-dispersion molecular gas is seen to be approximately a third of that seen in the ionized gas, i.e. 2.5 kpc (Fig. 4). We investigate this further by zooming into the central regions in the following subsection.

5.3 Kinematics on the central scale

Following the identification of non-gravitational gas motions identified in the central region of J1316+1753, presented in the previous sub-section, we were motivated to map the ionized and molecular gas

⁶Specifically, the Voronoi bins 0, 1, 2, 13, 17, 19 (see Fig. A1).

properties in more detail in this region. To do this, we mapped ionized and molecular gas kinematics across the central, 10×10 kpc region (see Section 2) using smaller spatial bins, and with a SNR cut of ≥ 3 . For the central-scale maps, we performed the emission-line profile fits, in spatial regions of 0.2×0.2 arcsec, i.e. in individual spaxels for the MUSE data and in bins of 4×4 pixels for the ALMA data to have a comparable resolution as MUSE. The results are described below and are illustrated in Fig. 5 for ionized gas and in Fig. 6 for the molecular gas. As discussed in Section 4.3, we lack the sensitivity to fit the stellar continuum with finer binning; however, in Section 6.2.2 we discuss the remarkable alignment between the jets and stellar emission and kinematics in this region using the larger spatial bins.

5.3.1 Ionized gas around the jet

In Fig. 5, we show maps of SNR, W_{80} , V_{50} , and n_e , along with some line profiles for the [O III] emission (see figure caption for detail). Contours of radio emission are overlaid, showing the relative location of the radio jets. Particularly striking is the high levels of velocitydispersion running perpendicular to the jet axis, apparent in the W_{80} maps shown in panel (b). The high velocity-dispersion in the ionized gas is further supported by the position–velocity (PV) diagram shown in panel (d), which was constructed by summing over a 0.6 arcsecwide pseudo-slit, along the jet axis. It is quite clear that these central regions show a broad velocity component with maximum velocities of around ± 1500 km s⁻¹.

Indeed, a visual inspection of the [O III] emission-line profile just north and just south of the jets [see panel (f) and (g), respectively] reveals a very broad component in these regions, with FWHM = $980 \pm 46 \text{ km s}^{-1}$ and $1310 \pm 18 \text{ km s}^{-1}$, respectively. North of the jet, this component is blueshifted by 100 km s^{-1} with respect to the local stellar velocity and south of the jet this component is redshifted by 140 km s^{-1} . We note that this shows the opposite red- and blue-shifts with respect to the galaxy rotation. This highest velocity-dispersion emission (with $W_{80} \ge 1000 \text{ km s}^{-1}$) starts from just behind the jet hot spots (highlighted with green dots in the W_{80} map and vertical dot–dot-dashed lines in the PV diagram).

All of these observations suggest outflowing, high velocitydispersion gas that originates from behind the jet hot spots and expands, in what appears to be a bi-cone structure. We trace the outer-edges of these cones by drawing straight dot–dashed lines, on the W_{80} map for spaxels along the axis perpendicular to the jet and identified with $W_{80} \ge 1000 \text{ km s}^{-1}$ to identify the morphology of the most extreme gas kinematics (see Section 5.2.2). These cones have opening angles of ~40°. To understand how this translates to the observations we made on the galaxy scales (Section 5.2), we overlaid them on to Fig. 4. Re-reassuringly, this gives a consistent picture because the highest velocity-dispersion gas seen on the galaxy-scale in Fig. 4 (i.e. out to 7.5 kpc each side of the nucleus), is located within these cones.

In panel (h), we show the electron density in the individual spaxels as estimated by using the ratio of the [S II] doublet (see Appendix B for more details and caveats). This reveals that the spaxels with the highest inferred electron density (i.e. $500-600 \text{ cm}^{-3}$) are co-spatial with the cones of highest velocity-dispersion ionized gas. We verified that this behaviour continues on to the galaxy scale by creating a map with electron density for each of the Voronoi bins and present the results in Appendix B. The spaxels outside the cones of highest velocity-dispersion regions have more modest electron densities (\leq

150 cm⁻³). We discuss the implications of these observations in Section 6.1.

Another key observation of the [O III] emission in the central regions, presented in Fig. 5, is that the highest SNR emission is concentrated in two regions; one around each of the two radio jet hot spots. A visual inspection of the emission-line profiles demonstrates that this is due to two narrow emission-line components that are particularly strong in each of these locations. These narrow emission-line components, which can be seen in the emission-line profiles shown in panel (e) and panel (g) of Fig. 5, are separated in velocity by 441 ± 14 km s⁻¹. Indeed, the V₅₀ map, shown in panel (c) of Fig. 5, highlights this further; we observe a strong velocity gradient *directly* along the jet axis. We note that this is offset by 20° from the galaxy kinematic axis and we have already shown that the gas in these regions is not dominated by gravitational motions (Section 5.2.2). The PV diagram in panel (d) of Fig. 5 shows that these two strong narrow emission-line components, which are highlighted with horizontal arrows on the panel, are emanating with velocities of $+206 \pm 15$ and $-235 \pm 1.3 \text{ km s}^{-1}$ from the location of the two jet hot spots. These jet hotspot locations are highlighted with vertical lines in the panel. We discuss these ionised gas velocity components in the context of jet-ISM interactions in Section 6.1.

5.3.2 Molecular gas around the jet

In Fig. 6, we present a similar set of figures for the CO(3–2) emission in the vicinity of the radio jets, as we did for the [O III] emission in Fig. 5, namely, maps of SNR, W_{80} , and V_{50} , along with some line-profiles for the CO(3–2) emission (see figure caption for detail). We observe the presence of molecular gas with high velocity-dispersion perpendicular to the jet axis [Fig. 6; panel (b)]. A visual inspection of the emission-line profile of this region [see panel (g)], shows that there is a broad CO(3–2) emission-line component (FWHM of $404 \pm 10 \text{ km s}^{-1}$) responsible for the high W_{80} values seen in this location. Furthermore, panel (h) shows a PV diagram extracted along a 0.6 arcsec slit perpendicular to the jet axis, which further confirms a region of high velocity-dispersion along this axis.

The disturbed molecular gas is located within the same cone identified in the ionized gas phase [see dot-dashed lines on panel (b) of Fig. 5], but, this time, only in the northern side of the jet. Furthermore, the disturbance appears to be less extreme. Specifically, the FWHM of the broad [O III] emission-line component in the same region is $3 \times$ higher than that observed in CO(3–2) at the same location. Additionally, the disturbed molecular gas is seen to extend to 2.5 kpc, which is only a third the extension of the disturbed ionized gas.

In the PV diagram of CO(3–2), created using a 0.6 arcsec pseudoslit running along the jet axis (panel (d) in Fig. 6), we observe a small velocity jump of around 100 km s⁻¹ at the location of the northern radio hotspot (highlighted with an arrow in the figure). Further evidence for irregular kinematics in this region comes from a visual inspection of the emission-line profile [see panel (f) in Fig. 6] which shows that the profile is split into two components separated by 169 \pm 9 km s⁻¹. Finally, there is some evidence for a deficit of CO(3– 2) emission close to the northern hotspot of the jet, as indicated by a black arrow in the figure. From the PV diagram, we estimate the SNR of the CO flux to be 11–14 times higher before and after this cavity and about 90 per cent less flux is seen in the cavity as compared to the surroundings. These observations coupled with the strong



Figure 6. Kinematic analysis of the molecular gas in the region around the jet, using maps of the non-parametric values of the CO(3-2) emission-line fits and position velocity (PV) diagrams around the CO(3-2) emission line (see Section 5.3.2). The black contours in each of the maps represent the 6 GHz radio emission at levels of [32,16,4] RMS_{radio}. **Panel (a)** : Peak-SNR map; **panel (b)** : W_{80} map; **panel (c)** : V_{50} map; **panel (d)** : PV diagram extracted along the jet axis (see red dashed axes in panel (a), with the position-offset along *x*-axis and velocity-offset along *y*-axis, and the vertical lines indicate the hotspots of the radio jets; **panels (e) and (f)** : CO(3-2) line profiles from the locations of the jet hotspots (Voronoi bin 4 and 19); **panel (g)** : CO(3-2) line profile from region of the broad disturbed gas just above the jet (Voronoi bin 16) and **panel (h)** : PV diagram extracted along an axis perpendicular to the jets (see yellow dashed axis in panel a). The components in the line-fit plots are as described in Fig. 3. Panel (b) shows the presence of disturbed molecular gas in a localized region perpendicular to the jet (Voronoi bin 16). The PV diagram in panel (d) reveals a break in the velocity profile at the north-west jet hotspot (HR:C, highlighted with an arrow).

narrow emission-line components seen in the ionized gas profile, serve as evidence of possible jet–ISM interactions happening in this region.

6 DISCUSSION

We have presented new MUSE and ALMA data for a type-2, z = 0.15 quasar, J1316+1753, selected from the Quasar Feedback Survey (Fig. 1). Based on the previous radio data, this source contains low-power ($P_{\text{jet}} = 10^{44} \text{ erg s}^{-1}$) ~1 kpc radio jets. These jets are confined within the galaxy disc and inclined at $\leq 5^{\circ}$ with respect to the plane of the galaxy disc, with the south-east jet (HR:B) as the approaching side (see Section 3.2 and Fig. 2). We summarize our key observations on the stellar, warm ionized gas and cold molecular gas kinematics with a schematic representation in Fig. 7. This is presented both from the line-of-sight point of view (left panel) and an edgeon view (right panel). Below, we discuss the observational features that appear to be tracing the interaction of the jets with the host galaxy's ISM (Section 6.1), before discussing the possible evidence for feedback effects on the host galaxy (Section 6.2) and then discussing the results in the context of the overall AGN population (Section 6.3).

6.1 Turbulent outflows and jet-ISM interactions

Here, we describe how our observations are consistent with the low power jets, which are highly inclined into the galaxy disc, are interacting with the ISM and driving large-scale turbulent outflows.

We see evidence for jet-ISM interactions on the multiphase gas at the jet termini (highlighted with blue/red regions in Fig. 7). Strong velocity enhancements in the warm ionized gas are seen, with two high velocity offset ionized gas components, separated by 441 km s⁻¹, that are brightest in the regions of the jet termini and appear to propagate away along the jet axis (Fig. 5). In the molecular gas phase, $a - 100 \text{ km s}^{-1}$ change in the molecular gas velocity is observed just beyond the HR:C radio hotspot (Fig. 6). This is consistent with previous work that has found velocity jumps in multiple phases of gas at the locations of jet termini (e.g. Morganti et al. 2013; Riffel, Storchi-Bergmann & Riffel 2014; Tadhunter et al. 2014; Morganti et al. 2015; Ramos Almeida et al. 2017; Finlez et al. 2018; Jarvis et al. 2019; Husemann et al. 2019; Ruffa et al. 2019; Santoro et al. 2020; Morganti et al. 2021). Furthermore, our results confirm that the strongly double peaked [O III] emission-line profile, seen in the SDSS spectrum of J1316+1753, is not the result of the presence of a dual AGN despite it being such a candidate (Xu & Komossa 2009; Smith et al. 2010; Lyu & Liu 2016; Baron & Poznanski 2017). Indeed, studies have also shown that double-peaked emission-lines



Figure 7. A schematic view of the galaxy J1316+1753 to highlight the key observations identified in this work. Left: The line-of-sight view (where North is up and East is to the left). Right: Our inferred edge-on view. A legend shows the different symbols used to represent the main features of interest and an approximate scale bar is shown in the right-hand panel at the bottom. A detailed description of the features in this schematic is presented in Section 6.1.

could mostly be an indication of the existence of (jet-driven) galaxywide outflows (see Rosario et al. 2010; Kharb et al. 2015; Nevin et al. 2018; Kharb et al. 2019, 2021, Ramos Almeida et al. 2021).

We also observe tentative evidence for depleted CO (3–2) emitting gas at the location of the brightest jets (see Fig. 6). We indicate this with a green-dashed circle in the schematic in Fig. 7 (also see top-right panel, Fig. 6). This is the same region where we see a bright warm ionized gas component in the [O III] (Fig. 5). Possible explanations for this include molecular gas removal or an interaction between the radio jets and the ISM that suppresses CO emission (e.g. through excitation or dissociation), while boosting the emission of the warm ionized gas (e.g. Rosario et al. 2019; Shimizu et al. 2019). Obtaining more molecular gas tracers, such as multiple CO lines and rotational and vibrational H₂ lines would help to verify this observation and differentiate between these scenarios (e.g. Rigopoulou et al. 2002; Dasyra et al. 2016).

Further evidence for an impact of the jets on the host galaxy ISM comes from the very high velocity-dispersion ionized gas (W80 \geq 1000 km s⁻¹; i.e. $\sigma \sim$ 400 km s⁻¹) originating from just behind the jet hotspots, and outflowing in an apparent bi-cone, in a perpendicular direction as far as project distance of 7.5 kpc (see Figs 4 and 5). The regions of this high velocity-dispersion ionized gas are highlighted as the purple hatched regions in the schematic diagram in Fig. 7. This disturbed gas can be seen to be moving along the galaxy minor axis and due to the orientation of the galaxy is seen to be moving towards us (and hence blue-shifted) above the galaxy plane and moving away from us (and hence red-shifted) below the galaxy plane. We also observe that this turbulent gas has a three times higher electron density than the rest of the galaxy disc (see Fig. 5). Such high electron densities within outflowing components have also been seen in previous studies (e.g. Villar Martín et al. 2014; Perna et al. 2017; Mingozzi et al. 2019; Davies et al. 2020; Fluetsch et al. 2021). A possible explanation for this could be due to the compression of the gas expelled in the outflow (e.g. Bourne, Zubovas & Nayakshin 2015; Decataldo et al. 2019).

In the molecular phase, we also see enhanced velocity-dispersion in the same direction as the turbulent ionized gas, but only in the regions north of the jet (not indicated in Fig. 7 for simplicity) and extends to about a distance of 2.5 kpc. A possible explanation for the less extreme kinematics observed in CO, compared to [O III], could be that the low velocity gas has cooled more efficiently, as the higher velocity gas has a higher post-shock temperature (e.g. Costa, Sijacki & Haehnelt 2015). The densest material in the ISM will also naturally be driven to lower velocities, which would also result in a smaller spatial extent of the turbulent/outflowing cold molecular gas compared to the less-dense warm ionized gas (see Nayakshin & Zubovas 2012; Mukherjee et al. 2016, 2018a). Furthermore, it is interesting to see the asymmetry (i.e. unipolar nature) of the CO(3-2) cones. One of the possible reasons could be that the galaxy disc maybe optically thicker to the CO(3-2) emission, thus hiding the other side of the turbulence, or the molecular gas could itself be intrinsically asymmetric as predicted by simulations at higher redshifts (Gabor & Bournaud 2014) and observations at lower redshifts (Lutz et al. 2020).

Our results are in qualitative agreement with recent hydrodynamical simulations showing that when a radio jet propagates through an inhomogenous medium, its impact can go beyond the radiojet axis. This happens because as the jets progress, they launch outflows close to them which disturb the local velocity flows and kinematics. However, they also inflate a spherical bubble of jet plasma that can lift the multiphase gas and cause turbulence, which would naturally be collimated along the minor axis of the host galaxy disc (Sutherland & Bicknell 2007; Wagner & Bicknell 2011; Wagner, Bicknell & Umemura 2012; Mukherjee et al. 2016, 2018a; Talbot, Bourne & Sijacki 2021). This would mean the disturbed gas would be extended beyond the radio emission (see also Zovaro et al. 2019). Mukherjee et al. (2018a, b) further predict a stronger impact on the host-galaxy's ISM when low power jets are inclined into the plane of the galaxy disc and have only moderate power, which matches the characteristics of the jets in J1316+1753. Within these simulations,

as the jets flood through the inter-cloud galaxy disc, they cross and interact more with the inter-stellar medium due to their lower inclination. We thus attribute the turbulent, high velocity-dispersion gas being vented out perpendicular to the galaxy disc in J1316+1753 to be originating as a consequence of the jets propagating through the ISM. Our observations are less consistent with a model where a quasar-driven shocked wind propagates through the ISM and causes both the radio emission and the turbulence (e.g. Hwang et al. 2018).

6.2 Feedback effects on the host galaxy

While the effect of radio jets is well studied and established at the scales of galaxy haloes, particularly for massive early-type galaxies and in cluster environments (McNamara & Nulsen 2012; Hardcastle & Croston 2020), it is less clear to what extent galaxies can be directly affected by low power jets confined within the ISM. Recent simulations show that, while on one hand density enhancements near the jet axis can make gravitational collapse of clouds more probable, thereby supporting star formation; an overall increase in turbulent kinetic energy may also lower the star formation rate (Mandal et al. 2021). From an observational perspective, studies have shown how jets could potentially suppress or regulate the starformation in the host galaxy by introducing turbulence in the ISM or causing multi-phase outflows (e.g. Nesvadba et al. 2010, 2011; Alatalo et al. 2015). On the contrary, studies have also shown a jetinduced positive feedback being driven by the shocks from the jets causing the gas to compress and thus inducing star formation (e.g. Bicknell et al. 2000: Salomé, Salomé & Combes 2015: Lacy et al. 2017). Hence, a jet may cause both a positive and a negative feedback on its host galaxy. Since we observe significant jet-ISM interactions and outflowing material in our observations, we now investigate the possible feedback effects arising from these on the host galaxy.

6.2.1 Impact of the jets on the molecular star-forming gas

To understand the impact of the jets in J1316+1753, we first quantify how much energy has been available from the jets over the lifetime of the observed outflows, using order-of-magnitude estimates. We assume a single jet episode, since we lack any evidence of multiple jet episodes. Under the assumption that the turbulent outflowing ionized gas is launched by the jet, we use it's kinematics to estimate the lifetime. The outflow velocity in the ionized gas is estimated to be $V_{\text{out}} = 800 \text{ km s}^{-1}$, at a distance of $R_{\text{out}} = 7.5 \text{ kpc}$ (see Section 5.2.2).⁸ If we then estimate the outflow lifetime as

$$t_{\rm out} \approx R_{\rm out}/v_{\rm out},$$
 (4)

we obtain $t_{out} \sim 9$ Myr. Using the value of the jet power ($P_{jet} = 10^{44} \text{ erg s}^{-1}$) estimated in Section 3, we estimate the total energy from the jet over this period as

$$E_{\rm jet} = P_{\rm jet} \times t_{\rm out} \sim 3 \times 10^{58} \,\rm erg. \tag{5}$$

Next, we estimate the kinetic energy, $E_{\text{mol, gas}}$ in the molecular gas phase (traced via CO) of the galaxy disc following Nesvadba et al. (2010), i.e. using the relationship between molecular gas mass, $M_{\text{mol, gas}}$ and velocity-dispersion:

$$E_{\rm mol,gas} = 0.5 \times M_{\rm mol,gas} \times \sigma_{\rm mol,gas}^2.$$
(6)

⁸We note that if we use the properties of the CO gas instead of the [O III] we would obtain an estimated lifetime of $t_{\rm out} \sim 6$ Myr, which is sufficiently close to that derived from [O III] for our order-of-magnitude estimates.

We focus on the molecular gas phase since the molecular gas reservoirs play a determining role in the AGN-galaxy co-evolution, as it is this gas that is redistributed to promote the star-formation in the galaxy and also for fuelling the black hole growth in the AGN (Carilli & Walter 2013; Vito et al. 2014). To determine the molecular gas mass, we first convert our measured CO fluxes into a CO (3-2) emission-line luminosity ($L'_{CO(3-2)}$ [K km s⁻¹ pc⁻²]), following Solomon et al. 1997 and then into $L'_{CO(1-0)}$, by adopting two different values for the line brightness temperature ratio, $r_{31} = 0.5$ and $r_{31} = 1$ (following Weiss et al. 2007; Riechers et al. 2011; Bolatto, Wolfire & Leroy 2013; Sargent et al. 2014). The former value is more consistent with quiescent star-forming gas, while the latter is for disturbed gas affected by the AGN. Finally, the molecular mass was estimated by using two values of the H2-to-CO conversion factor of $\alpha_{CO} \approx 4~M_{\odot}/(K~km~s^{-1}~pc^{-2})$ and $\alpha_{CO} \approx 0.89~M_{\odot}/(K~km$ s⁻¹ pc⁻²). The choice of the former α_{CO} is justified in Jarvis et al. (2020) and is consistent with that found in the Milky Way and assumed for quiescient gas. The latter choice is for typical highredshift, high star-forming, quasar host galaxies, following (Solomon et al. 1987; Solomon & Vanden Bout 2005; Carilli & Walter 2013; Bolatto et al. 2013). Although uncertain, these conversion factor choices are sufficient for our order-of-magnitude estimates (see further discussion in Jarvis et al. 2020). If we focus on the pixels inside the 2.5 kpc region showing the disturbed outflowing molecular gas (i.e. pixels with $W_{80} \ge 400 \,\mathrm{km \, s^{-1}}$, shown in Fig. 6), the range of molecular mass in the broad component was estimated to be $\sim 3.1 \times 10^7 \, M_{\odot} - 2.8 \times 10^8 \, M_{\odot}$ (using the different conversion factors). Hence, this broad component is seen to consist ~ 1 per cent of the total CO(3-2) flux. Furthermore, using the measured velocity, $\sigma_{\rm mol, gas} = 230 \,\rm km \, s^{-1}$, the total kinetic energy was estimated to be $E_{\rm mol,gas} \sim 1.6 \times 10^{55} \, {\rm erg} - 1.4 \times 10^{56} \, {\rm erg}.$

Hence, with a jet power higher by two orders of magnitude than the molecular kinetic energy, the jets are sufficient to drive the observed outflow. However, comparing the outflowing molecular gas mass to the total molecular mass in the galaxy, which is estimated as $\sim 10^{10} \, M_{\odot}$ (see Jarvis et al. 2020), we conclude that only a very small fraction of the total molecular gas mass is residing in this turbulent, outflowing material. This suggests little *in situ* disruption to the overall molecular gas disc (also see Ramos Almeida et al. 2021).

We further explore the possible fate of the molecular gas by comparing this energy to the galaxy binding energy. We used the stellar mass estimate of $M_{\star} = 10^{11} \,\mathrm{M}_{\odot}$ from Jarvis et al. (2019) and the stellar velocity-dispersion of ~ 300 km s⁻¹ (computed as an average value for the central bins in Fig. 4). Hence, the binding energy of the galaxy is estimated as,⁹

$$E_{\star} = M_{\star} \times \sigma_{\star}^2 \approx 10^{59} \,\mathrm{erg.} \tag{7}$$

Comparing the jet power with the binding energy of the galaxy, we thus conclude that the current jet episode will be limited in its ability to completely unbind the molecular gas from the host galaxy. However, negative AGN feedback may still manifest in several forms (see Federrath & Klessen 2012; Costa et al. 2020; Mandal et al. 2021), for example: (i) by quickly ejecting the dense gas from only the nuclear regions, as soon as the AGN outburst begins; (ii) by gradually suppressing the halo gas accretion on to the host galaxy; (iii) by inducing turbulence that provide additional support against

 $^{^{9}}$ We note that if we estimate the binding energy following the universal relationship between stellar mass and binding energy presented in Shi et al. (2021) we obtain the same order-of-magnitude result.

the collapse, thereby strongly suppressing star formation. Indeed, recent jet-ISM simulations do not predict a complete unbinding of host galaxy star-forming material, but rather a suppression of star-formation at the global galaxy scales by a factor of a few due to the ablation and fragmentation of the clouds (Mukherjee et al. 2018b; Mandal et al. 2021).

6.2.2 Evidence for feedback from stellar kinematics

We now focus on the possible signatures of feedback on the stellar properties of J1316+1753. We can clearly see a bright elongated bulge-like structure in the stellar continuum emission (Fig. 2, left-hand panel). Following Section 4.2, we fitted a two-dimensional Gaussian to the surface brightness profile of this bulge-shape, and estimated a position angle of 130°, which is remarkably close to the position angle of the jet axis, at 120° (red dashed line in Figs 5 and 6). In addition, the regions with highest stellar dispersion (i.e. \sim 290–360 km s⁻¹) follow, almost exactly, the jet axis (see the left-bottom panel of Fig. 4 and the galaxy schematic Fig. 7). We discuss two possible explanations of these observations.

We have seen clear signatures of jet-ISM interactions. A compression of the ISM gas caused by the jets could therefore trigger episodes of star formation, i.e. positive feedback. Stars formed during feedback will initially have motions which are significantly perturbed from regular circular stellar kinematics, and hence they may contribute to increasing the stellar velocity-dispersion and could alter the stellar structure of the host galaxy. This could be due to either directly induced positive feedback by jets or outflows, or because of stars that have formed in outflowing/turbulent material (see Ishibashi et al. 2013; Zubovas et al. 2013; Dugan et al. 2014; Maiolino et al. 2017; Gallagher et al. 2019). This may explain the increased velocitydispersion along the jet axis from our observations. Furthermore, at the jet termini, we can see gas velocities of -235 ± 1.3 and 206 ± 15 km s⁻¹, which entails to the jets pushing the gas with this velocity (the velocities seen in the two regions of enhanced [O III] SNR, or regions of jet-ISM interactions in Fig. 5). Interestingly, this roughly matches the higher stellar velocity-dispersion seen in these regions (-268 ± 18 and 240 ± 14 km s⁻¹, see Fig. 4). Detailed analyses of the stellar populations found in the different stellar kinematic components would be required to further test this scenario.

Another possible reason for this high stellar dispersion could be due to the sudden removal of gas in the very central regions, thereby causing a decrease in the local gravitational potential that may lead to ballistic orbits of central stars (see van der Vlugt & Costa 2019), thereby causing an apparent increase in the stellar velocitydispersion. However, it is not immediately clear why this would cause enhanced stellar velocity-dispersion along the jet axis and detailed dynamical modelling (accounting for the relative mass in stars, dark matter and gas) would be required to test this scenario further.

6.3 Implications of our results for 'radio-quiet' AGN feedback

Jet–ISM interaction similar to the one reported here have been identified in several other radio-quiet AGN host galaxies. We compare our source to a list of 13 targets, compiled by Venturi et al. (2021), which show a similar observation of high velocity-dispersion gas in the perpendicular direction (Section 6.1). In Fig. 8, we demonstrate this comparison of J1316+1753 in the $L_{bol}-P_{jet}$



Figure 8. A comparison of jet power (P_{jet} ; scaled directly from 5 GHz radio luminosity) versus L_{Bol} for radio-quiet AGN-host galaxies from literature, which show high velocity-dispersion gas, perpendicular to radio jets (see Section 6.3). Although this effect has been seen numerous times in the Seyfert-luminosity regime, J1316+1753 target clearly lies in the quasar regime as identified using our definition ($L_{[O III]} \ge 10^{42.1}$ erg s⁻¹, dashed line). The dashed orange line traces the division between 'radio-quiet' and 'radio-loud' AGN from Xu et al. (1999).

plane. We derived the P_{jet} from the 5 GHz radio luminosities¹⁰ in literature, following equation (1). This approach is sufficient for our order-of-magnitude investigation of the parameter space in Fig. 8. We used the $L_{Bol}/L_{[O III]} = 3500$ ratio following Heckman et al. (2004) to obtain the L_{Bol} values from the $L_{[O III]}$ values in

literature.¹¹

It can be clearly seen that the target we study here has a higher bolometric luminosity than those typically identified to have these type of jet–ISM interactions. Although J1316+1753 is at the upper end of jet powers, with $P_{jet} = 10^{44} \text{ erg s}^{-1}$ compared to the comparison sample in Fig. 8 (which reach $P_{jet} < 10^{44} \text{ erg s}^{-1}$), our target still lies in the radio-quiet regime. However, in contrast to the other sources, this target is a quasar, with $L_{\rm [O III]} \ge 10^{42.1} \text{ erg s}^{-1}$ (shown as dashed vertical line in the figure). Therefore, we would like to stress that for quasars and in general if $L_{\rm Bol} >> P_{\rm jet}$, jets could still be present as strong sources of feedback and hence should not be ignored (see also Villar-Martín et al. 2017). Indeed with the increasing availability of deep, high-resolution imaging, an increasing number of systems are also showing the presence of jets (e.g. Jarvis et al. 2019; Pierce, Tadhunter & Morganti 2020; Villar Martín et al. 2021).

In summary, we have shown that for J1316+1753, low power jets are the dominant mechanism for the central AGN to have an impact on the host galaxy's ISM, even when the system is bolometrically pow-

¹⁰For three of the sources where 5 GHz values were not available, we used 1.4 GHz values and converted them using a simple power law and an average spectral index $\alpha = -0.7$ to calculate 5 GHz luminosities. References used for radio luminosities are Jarvis et al. (2019), Griffith et al. (1994), Gallimore et al. (2006), Bicay et al. (1995), Griffith et al. (1995), Disney & Wall (1977), Whiteoak (1970), and Wright & Otrupcek (1990).

¹¹We note that since we could not obtain the $L_{[O III]}$ values for two of the sources, they are excluded from Fig. 8. This does not affect our results since the two excluded sources have X-ray flux emission that might be rather associated to X-ray binaries than with accretion on a SMBH. References used for $L_{[O III]}$ are Jarvis et al. (2019), Muñoz Marín et al. (2007), Schmitt et al. (2003), and Whittle (1992).

erful. Interestingly, by comparing the hydrodynamical simulations of individual galaxies that invoke a jet (Mukherjee et al. 2018b; Mandal et al. 2021) with those that invoke a quasar-wind (Costa, Sijacki & Haehnelt 2014; Costa et al. 2020) suggests that both mechanisms may have a very similar impact on the host galaxy (also see Faucher-Giguère & Quataert 2012; Talbot et al. 2021); specifically, both localized enhanced star formation (due to compression of gas) and more global, gradual suppression of star formation. Understanding which of these modes dominate for quasar host galaxies, and how representative J1316+1753 is across the quasar population, will require further studies combining high-quality radio imaging with spatially resolved measurements of the multiphase gas. We plan such a study across the wider QFeedS sample, where it will become possible to investigate trends with jet power and jet inclination angle, which are two crucial parameters according to simulations (Mukherjee et al. 2018b). Further careful studies of the role of low power jets for AGN feedback is important because lower power jets are very common, compared to their more powerful counterparts, in high-mass galaxies (including quasars) even if it is harder to decouple their radio emission from that produced by star formation (Gürkan et al. 2019; Sabater et al. 2019; Macfarlane et al. 2021).

7 CONCLUSIONS

We present MUSE and ALMA data of J1316+1753, a luminous z = 0.15, type-2 quasar selected from the Quasar Feedback Survey (Jarvis et al. 2021). This target represents one of the most [O III] luminous sources from the survey ($L_{[O III]} = 10^{42.8} \text{ erg s}^{-1}$) and exhibits a broad [O III] emission-line profile (FWHM~1300 km s⁻¹; Fig. 1; Fig. 3). Radio imaging of this source reveals low-power radio jets ($P_{jet} \sim 10^{44} \text{ erg s}^{-1}$) that are compact, reaching only 1 kpc in projected distance from the core. Furthermore, the jets are inclined into the plane of the host galaxy disc (see Section 3 and Fig. 2).

Our data enables us to map the stellar kinematics (traced with stellar absorption features), warm ionized gas properties (traced with optical emission-lines) and the cold molecular gas properties (traced with the CO(3–2) emission-line; see Fig. 3). On galaxy scales, both the molecular gas and ionised gas broadly follow the stellar gravitational motions (Fig. 4). However, across the central few kiloparsecs, both gas phases reveal high velocity non-gravitational motions and we observe evidence of jet-induced feedback. Specifically:

1. **Jet-ISM interactions**: We observe two bright and high velocity offset ionized gas components (separated by 441 km s⁻¹) concentrated at the positions of the jet hot spots and that appear to propagate away from the jets, along the jet axis (see Fig. 5). Furthermore, a -100 km s⁻¹ change in the molecular gas velocity is observed just beyond the brighter radio hotspot, with tentative evidence for depleted CO (3–2) emitting gas at the same location (see Fig. 6).

2. Multi-phase turbulent gas, driven perpendicular to the galaxy: Ionized gas with very high velocity-dispersion (i.e. $W_{80} = 1000 - 1300 \text{ km s}^{-1}$) is seen to propagate outwards in a bicone from just behind the radio hotspots, travelling perpendicular to the galaxy disc and seen in projection as extending to at least 7.5 kpc from the nucleus (see Figs 4 and 5). The highest inferred electron densities of the ionized gas are found within these bi-cones (as inferred from the [S II] doublet; Fig. 5). This turbulent gas is also seen in the molecular gas phase. However, it is 3 times less extended (and only in one direction) with 3 times lower velocity-dispersion (i.e. $W_{80} \sim 400 \text{ km s}^{-1}$) compared to ionized gas phase (see Figs 4 and 6).

3. Strong spatial connection between the jets and stellar properties: We see a close alignment of the position angle of the stellar bulge with the radio-jet axis ($< 5^{\circ}$; see Fig. 2, left-hand panel). Furthermore, the regions with the highest stellar velocity-dispersion (i.e. σ_{\star} are seen to be lagging behind the jets, following the jet axis (see Fig. 4, bottom left panel).

Our observations provide strong evidence for low power radio jets, inclined into the galaxy disc, having a direct impact on the multi-phase ISM inside the host galaxy of this type-2 quasar. Our observations are qualitatively consistent with the simultaneous positive and negative feedback effects observed in hydrodynamics simulations of jet-ISM interactions (Mukherjee et al. 2018a, b; Talbot et al. 2021; Mandal et al. 2021). Specifically: (1) as the inclined, lower power jets move through the galaxy, they strongly interact with the ISM; (2) highly turbulent material is stripped and escapes above and below the galaxy disc, removing gas from the host galaxy and; (3) as the jets propagate, they compress the gas in the disc also causing new stars to form in these regions, contributing to the formation of the stellar bulge (see Section 6.2.2). Although we only have indirect evidence, this final factor may be the cause of the high velocity-dispersion stars located behind the jets we have observed.

Interestingly, the same qualitative behaviour of outflows and positive feedback in the inner few kiloparsecs of galaxies is predicted by simulations invoking quasar-driven disc winds (Costa et al. 2020). Whilst jets have been seen to be dominant in lower power AGN (See Fig. 8 and Section 6.3), our observations reveal that jets can be the dominant feedback mechanism, even for this bolometrically luminous source, i.e. for an AGN currently in a 'quasar mode'. To understand the relative role and impact of jets and winds in all quasars requires a similar multiwavelength study of the wider population.

ACKNOWLEDGEMENTS

The authors would like to thank the referee for their comments which helped improved the quality of the manuscript. PK and SS acknowledge the support of the Department of Atomic Energy, Government of India, under the project 12-R&D-TFR-5.02-0700. DMA and ACE acknowledge support from STFC (ST/T000244/1). We would like to acknowledge the valuable insights from Dimitri Gadotti in helping us with the stellar spectral fits. This research made use of Astropy,¹² a community-developed core Python package for Astronomy (Astropy Collaboration et al. 2013, 2018).

This paper makes use of the following ALMA data: ADS/JAO.ALMA#2018.1.01767.S. ALMA is a partnership of ESO (representing its member states), NSF (USA) and NINS (Japan), together with NRC (Canada), MOST and ASIAA (Taiwan), and KASI (Republic of Korea), in cooperation with the Republic of Chile. The Joint ALMA Observatory is operated by ESO, AUI/NRAO and NAOJ.

Based on data obtained under ESO Proposal ID: 0103.B-0071.

DATA AVAILABILITY

The MUSE and ALMA data presented in this analysis were accessed from the ESO archive (http://archive.eso.org/scienceportal/home) under the proposal ids: 0103.B-0071 and 2018.1.01767.S, respectively. The VLA image used in this work is available at Newcastle University's data repository (https://data.ncl.ac.uk) and can also be accessed through our Quasar Feedback Survey website.

¹²http://www.astropy.org

REFERENCES

- Akaike H., 1974, IEEE Trans. Autom. Control, 19, 716
- Alatalo K. et al., 2015, ApJ, 798, 31
- Alexander D. M., Hickox R. C., 2012, New Astron. Rev., 56, 93
- Arentsen A. et al., 2019, A&A, 627, A138
- Astropy Collaboration et al., 2013, A&A, 558, A33
- Astropy Collaboration et al., 2018, AJ, 156, 123
- Bacon R. et al., 2010, in McLean I. S., Ramsay S. K., Takami H., eds, SPIE Conf. Ser. Vol. 7735, Ground-based and Airborne Instrumentation for Astronomy III. SPIE, Bellingham, p. 773508
- Bacon R. et al., 2017, A&A, 608, A1
- Bacon R., Piqueras L., Conseil S., Richard J., Shepherd M., 2016, MPDAF: MUSE Python Data Analysis Framework. Astrophysics Source Code Library, ascl:1611.003, https://git-cral.univ-lyon1.fr/MUSE/mpdaf
- Baron D., Netzer H., 2019, MNRAS, 486, 4290
- Baron D., Poznanski D., 2017, MNRAS, 465, 4530
- Benson A. J., Frenk C. S., Baugh C. M., Cole S., Lacey C. G., 2003, MNRAS, 343, 679
- Best P. N., Heckman T. M., 2012, MNRAS, 421, 1569
- Bicay M. D., Kojoian G., Seal J., Dickinson D. F., Malkan M. A., 1995, ApJS, 98, 369
- Bicknell G. V., Sutherland R. S., van Breugel W. J. M., Dopita M. A., Dey A., Miley G. K., 2000, ApJ, 540, 678
- Binney J., Tabor G., 1995, MNRAS, 276, 663
- Bittner A. et al., 2019, A&A, 628, A117
- Bittner A. et al., 2020, A&A, 643, A65
- Blandford R. D., Königl A., 1979, ApJ, 232, 34
- Bluck A. F. L. et al., 2020, MNRAS, 499, 230
- Bolatto A. D., Wolfire M., Leroy A. K., 2013, ARA&A, 51, 207
- Booth C. M., Schaye J., 2010, MNRAS, 405, L1
- Boroson T. A., Green R. F., 1992, ApJS, 80, 109
- Bourne M. A., Zubovas K., Nayakshin S., 2015, MNRAS, 453, 1829
- Bower R. G., Benson A. J., Crain R. A., 2012, MNRAS, 422, 2816
- Bower R. G., Benson A. J., Malbon R., Helly J. C., Frenk C. S., Baugh C. M., Cole S., Lacey C. G., 2006, MNRAS, 370, 645
- Bradley L. et al., 2020, Astropy/photutils: 1.0.0, Zenodo
- Cappellari M., 2017, MNRAS, 466, 798
- Cappellari M., Copin Y., 2003, MNRAS, 342, 345
- Cappellari M., Emsellem E., 2004, PASP, 116, 138
- Carilli C. L., Walter F., 2013, ARA&A, 51, 105
- Cattaneo A., Best P. N., 2009, MNRAS, 395, 518
- Choi E., Somerville R. S., Ostriker J. P., Naab T., Hirschmann M., 2018, ApJ, 866, 91
- Cicone C. et al., October, 2021, A&A, 654, 9
- Cicone C., Brusa M., Ramos Almeida C., Cresci G., Husemann B., Mainieri V., 2018, Nature Astron., 2, 176
- Ciotti L., Ostriker J. P., 1997, ApJ, 487, L105
- Circosta C. et al., 2021, A&A, 646, A96
- Combes F. et al., 2014, A&A, 565, A97
- Costa T., Pakmor R., Springel V., 2020, MNRAS, 497, 5229
- Costa T., Rosdahl J., Sijacki D., Haehnelt M. G., 2018, MNRAS, 479, 2079
- Costa T., Sijacki D., Haehnelt M. G., 2014, MNRAS, 444, 2355
- Costa T., Sijacki D., Haehnelt M. G., 2015, MNRAS, 448, L30
- Cresci G., Maiolino R., 2018, Nature Astron., 2, 179
- Dasyra K. M., Combes F., Oosterloo T., Oonk J. B. R., Morganti R., Salomé P., Vlahakis N., 2016, A&A, 595, L7
- Davies R. et al., 2020, MNRAS, 498, 4150
- Decataldo D., Pallottini A., Ferrara A., Vallini L., Gallerani S., 2019, MNRAS, 487, 3377
- Dimitrijević M. S., Popović L. Č., Kovačević J., Dačić M., Ilić D., 2007, MNRAS, 374, 1181
- Disney M. J., Wall J. V., 1977, MNRAS, 179, 235
- do Nascimento J. C. et al., 2019, MNRAS, 486, 5075
- Dugan Z., Bryan S., Gaibler V., Silk J., Haas M., 2014, ApJ, 796, 113
- Fabian A. C., 2012, ARA&A, 50, 455
- Falcón-Barroso J. et al., 2006, MNRAS, 369, 529
- Faucher-Giguère C.-A., Quataert E., 2012, MNRAS, 425, 605

Federrath C., Klessen R. S., 2012, ApJ, 761, 156

- Finlez C., Nagar N. M., Storchi-Bergmann T., Schnorr-Müller A., Riffel R. A., Lena D., Mundell C. G., Elvis M. S., 2018, MNRAS, 479, 3892
- Fluetsch A. et al., 2021, MNRAS, 505, 5753
- Fragile P. C., Anninos P., Croft S., Lacy M., Witry J. W. L., 2017, ApJ, 850, 171
- Gabor J. M., Bournaud F., 2014, MNRAS, 441, 1615
- Gadotti D. A. et al., 2020, A&A, 643, A14
- Gallagher R., Maiolino R., Belfiore F., Drory N., Riffel R., Riffel R. A., 2019, MNRAS, 485, 3409
- Gallimore J. F., Axon D. J., O'Dea C. P., Baum S. A., Pedlar A., 2006, AJ, 132, 546
- Gaspari M., Brighenti F., D'Ercole A., Melioli C., 2011, MNRAS, 415, 1549
- Ghisellini G., Padovani P., Celotti A., Maraschi L., 1993, ApJ, 407, 65
- Gonneau A. et al., 2020, A&A, 634, A133
- Granato G. L., De Zotti G., Silva L., Bressan A., Danese L., 2004, ApJ, 600, 580
- Griffith M. R., Wright A. E., Burke B. F., Ekers R. D., 1994, ApJS, 90, 179
- Griffith M. R., Wright A. E., Burke B. F., Ekers R. D., 1995, ApJS, 97, 347
- Gürkan G. et al., 2019, A&A, 622, A11
- Hardcastle M. J., Croston J. H., 2020, New A Rev., 88, 101539
- Harrison C. M., 2017, Nature Astron., 1, 0165
- Harrison C. M., Alexander D. M., Mullaney J. R., Swinbank A. M., 2014, MNRAS, 441, 3306
- Harrison C. M., Costa T., Tadhunter C. N., Flütsch A., Kakkad D., Perna M., Vietri G., 2018, Nature Astron., 2, 198
- Harrison C. M., Thomson A. P., Alexander D. M., Bauer F. E., Edge A. C., Hogan M. T., Mullaney J. R., Swinbank A. M., 2015, ApJ, 800, 45
- Heckman T. M., Kauffmann G., Brinchmann J., Charlot S., Tremonti C., White S. D. M., 2004, ApJ, 613, 109
- Heckman T. M., Miley G. K., van Breugel W. J. M., Butcher H. R., 1981, ApJ, 247, 403
- Henriques B. M. B., White S. D. M., Thomas P. A., Angulo R., Guo Q., Lemson G., Springel V., Overzier R., 2015, MNRAS, 451, 2663

Downloaded from https://academic.oup.com/mnras/article/512/2/1608/6515973 by Liverpool John Moores University user on 15 June 2023

- Hirschmann M., Dolag K., Saro A., Bachmann L., Borgani S., Burkert A., 2014, MNRAS, 442, 2304
- Hopkins P. F., Hernquist L., Cox T. J., Di Matteo T., Robertson B., Springel V., 2006, ApJS, 163, 1
- Husemann B. et al., 2019, ApJ, 879, 75
- Husemann B., Scharwächter J., Bennert V. N., Mainieri V., Woo J. H., Kakkad D., 2016, A&A, 594, A44
- Hwang H.-C., Zakamska N. L., Alexandroff R. M., Hamann F., Greene J. E., Perrotta S., Richards G. T., 2018, MNRAS, 477, 830
- Ishibashi W., Fabian A. C., Canning R. E. A., 2013, MNRAS, 431, 2350
- Ishibashi W., Fabian A. C., Maiolino R., 2018, MNRAS, 476, 512
- Jarvis M. E. et al., 2019, MNRAS, 485, 2710
- Jarvis M. E. et al., 2020, MNRAS, 498, 1560
- Jarvis M. E. et al., 2021, MNRAS, 503, 1780
- Kakkad D. et al., 2017, MNRAS, 468, 4205
- Keel W. C. et al., 2012, AJ, 144, 66
- Keel W. C. et al., 2019, MNRAS, 483, 4847
- Kharb P., Das M., Paragi Z., Subramanian S., Chitta L. P., 2015, ApJ, 799, 161
- Kharb P., Shastri P., 2004, A&A, 425, 825
- Kharb P., Subramanian S., Das M., Vaddi S., Paragi Z., 2021, ApJ, 919, 108
- Kharb P., Vaddi S., Sebastian B., Subramanian S., Das M., Paragi Z., 2019, ApJ, 871, 249
- King A., Pounds K., 2015, ARA&A, 53, 115
- Kormendy J., Ho L. C., 2013, ARA&A, 51, 511
- Lacy M., Croft S., Fragile C., Wood S., Nyland K., 2017, ApJ, 838, 146
- Lansbury G. B., Jarvis M. E., Harrison C. M., Alexander D. M., Del Moro A., Edge A. C., Mullaney J. R., Thomson A. P., 2018, ApJ, 856, L1

Liu G., Zakamska N. L., Greene J. E., Nesvadba N. P. H., Liu X., 2013,

MNRAS 512, 1608-1628 (2022)

Liao M., Gu M., 2020, MNRAS, 491, 92 Lintott C. J. et al., 2009, MNRAS, 399, 129

Lutz D. et al., 2020, A&A, 633, A134

Lyu Y., Liu X., 2016, MNRAS, 463, 24

MNRAS, 436, 2576

- Macfarlane C. et al., 2021, MNRAS, 506, 5888
- Maiolino R. et al., 2017, Nature, 544, 202
- Mandal A., Mukherjee D., Federrath C., Nesvadba N. P. H., Bicknell G. V., Wagner A. Y., Meenakshi M., 2021, MNRAS, 508, 4738
- McCarthy I. G. et al., 2010, MNRAS, 406, 822
- McNamara B. R., Nulsen P. E. J., 2012, New J. Phys., 14, 055023
- Merloni A., Heinz S., 2007, MNRAS, 381, 589
- Mingozzi M. et al., 2019, A&A, 622, A146
- Molyneux S. J., Harrison C. M., Jarvis M. E., 2019, A&A, 631, A132
- Morganti R., Frieswijk W., Oonk R. J. B., Oosterloo T., Tadhunter C., 2013, A&A, 552, L4
- Morganti R., Oosterloo T., Oonk J. B. R., Frieswijk W., Tadhunter C., 2015, A&A, 580, A1
- Morganti R., Oosterloo T., Tadhunter C. N., 2005, in Braun R., ed., ASP Conf. Ser. Vol. 331, Extra-Planar Gas. Astron. Soc. Pac., San Francisco, p. 361
- Morganti R., Oosterloo T., Tadhunter C., Bernhard E. P., Oonk J. B. R., 2021, A&A, 656, 14
- Mukherjee D., Bicknell G. V., Sutherland R., Wagner A., 2016, MNRAS, 461, 967
- Mukherjee D., Bicknell G. V., Wagner A. Y., Sutherland R. S., Silk J., 2018b, MNRAS, 479, 5544
- Mukherjee D., Wagner A. Y., Bicknell G. V., Morganti R., Oosterloo T., Nesvadba N., Sutherland R. S., 2018a, MNRAS, 476, 80
- Mullaney J. R., Alexander D. M., Fine S., Goulding A. D., Harrison C. M., Hickox R. C., 2013, MNRAS, 433, 622
- Muñoz Marín V. M., González Delgado R. M., Schmitt H. R., Cid Fernandes R., Pérez E., Storchi-Bergmann T., Heckman T., Leitherer C., 2007, AJ, 134, 648
- Nayakshin S., Zubovas K., 2012, MNRAS, 427, 372
- Nesvadba N. P. H. et al., 2010, A&A, 521, A65
- Nesvadba N. P. H., Boulanger F., Lehnert M. D., Guillard P., Salome P., 2011, A&A, 536, L5
- Nesvadba N. P. H., Lehnert M. D., De Breuck C., Gilbert A. M., van Breugel W., 2008, A&A, 491, 407
- Nesvadba N. P. H., Lehnert M. D., Eisenhauer F., Gilbert A., Tecza M., Abuter R., 2006, ApJ, 650, 693
- Nevin R., Comerford J. M., Müller-Sánchez F., Barrows R., Cooper M. C., 2018, MNRAS, 473, 2160
- Orr M. J. L., Browne I. W. A., 1982, MNRAS, 200, 1067
- Osterbrock D. E., Ferland G. J., 2006, Astrophysics of Gaseous Nebulae and Active Galactic Nuclei, University Science Books, Sausalito, CA, United States
- Perna M. et al., 2018, A&A, 619, A90
- Perna M., Lanzuisi G., Brusa M., Cresci G., Mignoli M., 2017, A&A, 606, A96
- Pierce J. C. S., Tadhunter C. N., Morganti R., 2020, MNRAS, 494, 2053
- Pinna F. et al., 2019, A&A, 623, A19
- Ramos Almeida C., Piqueras López J., Villar-Martín M., Bessiere P. S., 2017, MNRAS, 470, 964
- Ramos Almeida C. et al. 2021, preprint (arXiv:2111.13578)
- Riechers D. A. et al., 2011, ApJ, 739, L32
- Riffel R. A., Storchi-Bergmann T., Riffel R., 2014, ApJ, 780, L24
- Rigopoulou D., Kunze D., Lutz D., Genzel R., Moorwood A. F. M., 2002, A&A, 389, 374
- Rosario D. J. et al., 2018, MNRAS, 473, 5658
- Rosario D. J., Shields G. A., Taylor G. B., Salviander S., Smith K. L., 2010, ApJ, 716, 131
- Rosario D. J., Togi A., Burtscher L., Davies R. I., Shimizu T. T., Lutz D., 2019, ApJ, 875, L8
- Rose M., Tadhunter C., Ramos Almeida C., Rodríguez Zaurín J., Santoro F., Spence R., 2018, MNRAS, 474, 128
- Roy N. et al., 2021, ApJ, 919, 145
- Ruffa I. et al., 2019, MNRAS, 484, 4239
- Rupke D. S., Veilleux S., Sanders D. B., 2005, ApJS, 160, 115
- Sabater J. et al., 2019, A&A, 622, A17
- Saikia D. J., Kapahi V. K., 1982, Bull. Astron. Soc. India, 10, 42
- Salomé Q., Salomé P., Combes F., 2015, A&A, 574, A34

- Santoro F., Tadhunter C., Baron D., Morganti R., Holt J., 2020, A&A, 644, A54
- Sargent M. T. et al., 2014, ApJ, 793, 19
- Schaye J. et al., 2015, MNRAS, 446, 521
- Schmitt H. R., Donley J. L., Antonucci R. R. J., Hutchings J. B., Kinney A. L., 2003, ApJS, 148, 327
- Scholtz J. et al., 2021, MNRAS, 505, 5469
- Shi Y., Yu X., Mao S., Gu Q., Xia X., Chen Y., 2021, MNRAS, 507, 2423
- Shih H.-Y., Rupke D. S. N., 2010, ApJ, 724, 1430
- Shimizu T. T. et al., 2019, MNRAS, 490, 5860
- Silk J., Rees M. J., 1998, A&A, 331, L1
- Smith K. L., Shields G. A., Bonning E. W., McMullen C. C., Rosario D. J., Salviander S., 2010, ApJ, 716, 866
- Solomon P. M., Downes D., Radford S. J. E., Barrett J. W., 1997, ApJ, 478, 144
- Solomon P. M., Rivolo A. R., Barrett J., Yahil A., 1987, ApJ, 319, 730
- Solomon P. M., Vanden Bout P. A., 2005, ARA&A, 43, 677
- Somalwar J., Johnson S. D., Stern J., Goulding A. D., Greene J. E., Zakamska N. L., Alexandroff R. M., Chen H.-W., 2020, ApJ, 890, L28
- Stockton A., 1976, ApJ, 205, L113
- Sun A.-L., Greene J. E., Zakamska N. L., Nesvadba N. P. H., 2014, ApJ, 790, 160
- Sutherland R. S., Bicknell G. V., 2007, ApJS, 173, 37
- Tadhunter C., Morganti R., Rose M., Oonk J. B. R., Oosterloo T., 2014, Nature, 511, 440
- Talbot R. Y., Bourne M. A., Sijacki D., 2021, MNRAS, 504, 3619
- Taylor P., Kobayashi C., 2015, MNRAS, 448, 1835
- Tully R. B., Fisher J. R., 1977, A&A, 500, 105
- Ulvestad J. S., Wrobel J. M., Roy A. L., Wilson A. S., Falcke H., Krichbaum T. P., 1999, ApJ, 517, L81
- Urry C. M., Padovani P., 1995, PASP, 107, 803
- van der Vlugt D., Costa T., 2019, MNRAS, 490, 4918
- Veilleux S., Cecil G., Bland-Hawthorn J., 2005, ARA&A, 43, 769
- Veilleux S., Kim D. C., Sanders D. B., Mazzarella J. M., Soifer B. T., 1995, ApJS, 98, 171
- Veilleux S., Maiolino R., Bolatto A. D., Aalto S., 2020, A&A Rev., 28, 2
- Venturi G. et al., 2021, A&A, 648, A17
- Villar Martín M., Emonts B. H. C., Cabrera Lavers A., Bellocchi E., Alonso Herrero A., Humphrey A., Dall'Agnol de Oliveira B., Storchi-Bergmann T., 2021, A&A, 650, A84
- Villar Martín M., Emonts B., Humphrey A., Cabrera Lavers A., Binette L., 2014, MNRAS, 440, 3202
- Villar-Martín M. et al., 2017, MNRAS, 472, 4659
- Villar-Martín M., Arribas S., Emonts B., Humphrey A., Tadhunter C., Bessiere P., Cabrera Lavers A., Ramos Almeida C., 2016, MNRAS, 460, 130
- Virtanen P. et al., 2020, Nature Methods, 17, 261
- Vito F. et al., 2014, MNRAS, 441, 1059
- Vogelsberger M. et al., 2014, Nature, 509, 177
- Vrtilek J. M., 1985, ApJ, 294, 121
- Wagner A. Y., Bicknell G. V., 2011, ApJ, 728, 29
- Wagner A. Y., Bicknell G. V., Umemura M., 2012, ApJ, 757, 136
- Wang X., Loeb A., 2018, New A, 61, 95
- Weedman D. W., 1970, ApJ, 159, 405
- Weijmans A.-M. et al., 2014, MNRAS, 444, 3340
- Weilbacher P. M. et al., 2020, A&A, 641, A28
- Weilbacher P. M., Streicher O., Urrutia T., Pécontal-Rousset A., Jarno A., Bacon R., 2014, in Manset N., Forshay P., eds, ASP Conf. Ser. Vol. 485, Astronomical Data Analysis Software and Systems XXIII. Astron. Soc. Pac., San Francisco, p. 451
- Weiss A., Downes D., Walter F., Henkel C., 2007, in Baker A. J., Glenn J., Harris A. I., Mangum J. G., Yun M. S., eds, ASP Conf. Ser. Vol. 375, From Z-Machines to ALMA: (Sub)Millimeter Spectroscopy of Galaxies. Astron. Soc. Pac., San Francisco, p. 25
- Whiteoak J. B., 1970, Astrophys. Lett., 5, 29
- Whittle M., 1992, ApJ, 387, 109

- Wright A., Otrupcek R., 1990, PKS Catalog, Australia Telecsope National Facility, Commonwealth Scientific and Industrial Research Organization, Melbourne
- Wylezalek D., Morganti R., 2018, Nature Astron., 2, 181
- Xu C., Livio M., Baum S., 1999, AJ, 118, 1169
- Xu D., Komossa S., 2009, ApJ, 705, L20
- Yesuf H. M., Ho L. C., 2020, ApJ, 901, 42
- Zakamska N. L., Strauss M. A., Heckman T. M., Ivezić, Ž., Krolik J. H., 2004, AJ, 128, 1002
- Zovaro H. R. M., Sharp R., Nesvadba N. P. H., Bicknell G. V., Mukherjee D., Wagner A. Y., Groves B., Krishna S., 2019, MNRAS, 484, 3393
- Zubovas K., Nayakshin S., King A., Wilkinson M., 2013, MNRAS, 433, 3079

SUPPORTING INFORMATION

Supplementary data are available at MNRAS online.

suppl_data

Please note: Oxford University Press is not responsible for the content or functionality of any supporting materials supplied by the authors. Any queries (other than missing material) should be directed to the corresponding author for the article.

APPENDIX A: VORONOI BINNING MAP

In Fig. A1, we show a map with the spatial distribution and numbering system of the Voronoi bins, produced from the Voronoi tesselation routine, as explained in Section 4.3.



Figure A1. A map of the 41 Voronoi Bins obtained using a SNR of 5, as described in the Section 4.3. The same Voronoi mapping is used for the analysis of the MUSE and ALMA data at the galaxy-scale, as summarized in Sections 5.2.2 and 5.2.3, respectively. Five of the bins (no. 10, 14, 24, 29, 40) that were excluded from the analysis of the molecular gas due to low SNR are labelled in red. Also it can be seen here that the bins 38, 39, 40 correspond to the companion galaxy while the rest belong to the main galaxy as labelled in Fig. 2.

APPENDIX B: ELECTRON DENSITIES

We estimated the spatial variations in electron density by using the ratio of the [S II]λλ6716,6731 emission-line doublet, assuming a temperature of $T_e = 10^4$ K (Osterbrock & Ferland 2006). As described in Section 4.4, we fit the [S II] doublet with Gaussian components: three components for each line, as shown in the left-hand panel in Fig. B1 for two example bins. We note that the [S II] emission-line fits were fit independently for their flux, velocity, and linewidths, without any influence of the [O III] parameters obtained in Section 4.4. The right-hand panel of Fig. B1 shows a map of the estimated electron density values on the galaxy-scale using the Voronoi bins that were defined on the stellar continuum in Section 4.3. We further perform this analysis on the central-scale using individual MUSE spaxels, the results of which are shown in the bottom right-hand panel of Fig. 5. We note that the [S II] doublet is only sensitive to electron densities in the range $50 \le n_e \le 5000 \text{ cm}^{-3}$ (Osterbrock & Ferland 2006). Therefore, beyond this range, we indicate the electron densities are unconstrained in these regions through grey pixels in Fig. 5 and we discuss this limitation further below.

Interestingly, both the central-scale (10 kpc) and the galaxy-scale analyses (26 kpc) reveals that higher electron densities (i.e. \sim 500– 600cm^{-3}) are observed perpendicular to the radio jets. Furthermore, the regions of the highest inferred electron densities are remarkably co-spatial with the most disturbed ionized gas (see Section 5.3.1), whilst the spaxels outside of this cone exhibit more modest electron densities (i.e. $\sim 150 \text{ cm}^{-3}$). This has also been observed by other studies (see e.g. Mingozzi et al. 2019; Davies et al. 2020; Fluetsch et al. 2021). However, we urge caution in interpreting the absolute values of the electron densities inferred using this approach. For example, the [S II] doublet is limited in its diagnostic power for tracing electron densities and may give results that are about 1-2 orders lower in magnitude than the electron densities derived using trans-auroral [O II] and [S II] lines (for details, see Harrison et al. 2018). A recent study by Davies et al. (2020) also found significantly lower electron densities using this method as compared to the auroral and trans-auroral lines (see also Harrison et al. 2018; Rose et al. 2018; Baron & Netzer 2019; Shimizu et al. 2019).



Figure B1. A representation of the spatial distribution of electron densities (n_e) using the [S II] doublet method. Right-hand panel: A map colour-coded by n_e , estimated for each of the Voronoi bins, mapped over the entire galaxy. The contours are same as described in Fig. 2. The small inset box shows the central-scale, for which the map is presented in Fig. 5. Left-hand panels: Example [S II]-emission-line profiles and their fits are shown for regions of low (Bin 23) and high (Bin 16) electron densities. The data and fitting-components are displayed using the same convention as for Fig. 3.

This paper has been typeset from a TEX/LATEX file prepared by the author.